

# 

**Citation:** Russell NJ, Stöhr W, Plakkal N, Cook A, Berkley JA, Adhisivam B, et al. (2023) Patterns of antibiotic use, pathogens, and prediction of mortality in hospitalized neonates and young infants with sepsis: A global neonatal sepsis observational cohort study (NeoOBS). PLoS Med 20(6): e1004179. https://doi.org/10.1371/journal. pmed.1004179

Received: June 15, 2022

Accepted: January 19, 2023

Published: June 8, 2023

**Copyright:** © 2023 Russell et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its <u>Supporting</u> information files.

**Funding:** This study was funded by the Global Antibiotic Research and Development Partnership (GARDP), made possible by Bill & Melinda Gates Foundation; German Federal Ministry of Education and Research; German Federal Ministry of Health; Government of the Principality of Monaco; the Indian Council for Medical Research; Japanese Ministry of Health, Labour and Welfare; Netherlands Ministry of Health, Welfare and Sport; RESEARCH ARTICLE

Patterns of antibiotic use, pathogens, and prediction of mortality in hospitalized neonates and young infants with sepsis: A global neonatal sepsis observational cohort study (NeoOBS)

Neal J. Russell<sup>1‡\*</sup>, Wolfgang Stöhr<sup>2‡</sup>, Nishad Plakkal<sup>3</sup>, Aislinn Cook<sup>1</sup>, James A. Berkley<sup>4,5,6</sup>, Bethou Adhisivam<sup>3</sup>, Ramesh Agarwal<sup>7</sup>, Nawshad Uddin Ahmed<sup>8</sup>, Manica Balasegaram<sup>9</sup>, Daynia Ballot<sup>10</sup>, Adrie Bekker<sup>11</sup>, Eitan Naaman Berezin<sup>12</sup>, Davide Bilardi<sup>13</sup>, Suppawat Boonkasidecha<sup>14</sup>, Cristina G. Carvalheiro<sup>15</sup>, Neema Chami<sup>10</sup>, Suman Chaurasia<sup>16</sup>, Sara Chiurchiu<sup>17</sup>, Viviane Rinaldi Favarin Colas<sup>12</sup>, Simon Cousens<sup>18</sup>, Tim R. Cressey<sup>19</sup>, Ana Carolina Dantas de Assis<sup>15</sup>, Tran Minh Dien<sup>20</sup>, Yijun Ding<sup>20</sup>, Nguyen Trong Dung<sup>20</sup>, Han Dong<sup>21</sup>, Angela Dramowski<sup>11</sup>, Madhusudhan DS<sup>22</sup>, Ajay Dudeja<sup>23</sup>, Jinxing Feng<sup>24</sup>, Youri Glupczynski<sup>25</sup>, Srishti Goel<sup>23</sup>, Herman Goossens<sup>25</sup>, Doan Thi Huong Haa<sup>20</sup>, Mahmudul Islam Khan<sup>8</sup>, Tatiana Munera Huertas<sup>11</sup>, Mohammad Shahidul Islam<sup>8</sup>, Daniel Jarovsky<sup>12</sup>, Nathalie Khavessian<sup>9</sup>, Meera Khorana<sup>26</sup>, Angeliki Kontou<sup>27</sup>, Tomislav Kostyanev<sup>25</sup>, Premsak Laoyookhon<sup>14</sup>, Sorasak Lochindarat<sup>14</sup>, Mattias Larsson<sup>28</sup>, Maia De Luca<sup>17</sup>, Surbhi Malhotra-Kumar<sup>25</sup>, Nivedita Mondal<sup>3</sup>, Nitu Mundhra<sup>22</sup>, Philippa Musoke<sup>29</sup>, Marisa M. Mussi-Pinhata<sup>15</sup>, Ruchi Nanavati<sup>22</sup>, Firdose Nakwa<sup>10</sup>, Sushma Nangia<sup>23</sup>, Jolly Nankund<sup>30</sup>, Alessandra Nardone<sup>13</sup>, Borna Nyaoke<sup>9</sup>, Christina W. Obiero<sup>4,31</sup>, Maxensia Owor<sup>30</sup>, Wang Ping<sup>21</sup>, Kanchana Preedisripipat<sup>32</sup>, Shamim Qazi<sup>33</sup>, Lifeng Qi<sup>34</sup>, Tanusha Ramdin<sup>10,35</sup>, Amy Riddell<sup>1</sup>, Lorenza Romani<sup>17</sup>, Praewpan Roysuwan<sup>19</sup>, Robin Saggers<sup>35</sup>, Emmanuel Roilides<sup>36</sup>, Samir K. Saha<sup>8</sup>, Kosmas Sarafidis<sup>27</sup>, Valerie Tusubira<sup>29</sup>, Reenu Thomas<sup>10,37</sup>, Sithembiso Velaphi<sup>10</sup>, Tuba Vilken<sup>25</sup>, Xiaojiao Wang<sup>38</sup>, Yajuan Wang<sup>39</sup>, Yonghong Yang<sup>24</sup>, Liu Zunjie<sup>21</sup>, Sally Ellis<sup>9</sup>, Julia A. Bielicki<sup>1</sup>, A. Sarah Walker<sup>2</sup>, Paul T. Heath<sup>1</sup>, Mike Sharland<sup>1</sup>

1 Center for Neonatal and Paediatric Infection (CNPI), Institute of Infection & Immunity, St George's University of London, London, United Kingdom, 2 Medical Research Council Clinical Trials Unit at University College London, London, United Kingdom, 3 Department of Neonatology, Jawaharlal Institute of Postgraduate Medical Education & Research (JIPMER), Pondicherry, India, 4 Clinical Research Department, KEMRI-Wellcome Trust Research Programme, Kilifi, Kenya, 5 Centre for Tropical Medicine & Global Health, Nuffield Department of Medicine, University of Oxford, Oxford, United Kingdom, 6 The Childhood Acute Illness & Nutrition (CHAIN) Network, Nairobi, Kenya, 7 Newborn Division and WHO-CC, All India Institute of Medical Sciences, New Delhi, India, 8 Child Health Research Foundation (CHRF), Dhaka Shishu Hospital, Dhaka, Bangladesh, 9 Global Antibiotic Research and Development Partnership (GARDP), Geneva, Switzerland, 10 Department of Paediatrics and Child Health, School of Clinical Medicine, Faculty of Health Sciences, University of the Witwatersrand, Johannesburg, South Africa, 11 Department of Paediatrics and Child Health, Faculty of Medicine and Health Sciences, Stellenbosch University, South Africa, 12 Pediatric Infectious Diseases Unit, Santa Casa de São Paulo, São Paulo, Brazil, 13 Penta Foundation, Padova, Italy, 14 Queen Sirikit National Institute of Child Health, Bangkok, Thailand, 15 Department of Pediatrics, Ribeirão Preto Medical School, University of São Paulo, São Paulo, Brazil, 16 All India Institute of Medical Sciences, Department of Paediatrics, New Delhi, India, 17 Academic Hospital Paediatric Department, Bambino Gesù Children's Hospital, Rome, Italy, 18 Faculty of Epidemiology and Population Health, Department of Infectious Disease Epidemiology, London School of Hygiene & Tropical Medicine, London, United Kingdom, 19 PHPT/ IRD-MIVEGEC, Faculty of Associated Medical Sciences, Chiang Mai University, Chiang Mai, Thailand, 20 Vietnam National Children's Hospital, Hanoi, Vietnam and Surgical Intensive Care Unit, Vietnam National Children's Hospital, Hanoi, Vietnam, 21 Beijing Obstetrics and Gynecology Hospital, Capital Medical University, Beijing, China, 22 Neonatology Department, Seth GS Medical College and King Edward Memorial Hospital, Mumbai, India, 23 Department of Neonatology, Lady Hardinge Medical College and Kalawati Saran Children's Hospital, New Delhi, India, 24 Department of Neonatology, Shenzhen Children's Hospital,

South African Medical Research Council: UK Department of Health and Social Care (UK National Institute of Health Research and the Global Antimicrobial Resistance Innovation Fund -GAMRIF); UK Medical Research Council; Wellcome Trust. GARDP has also received core funding from the Leo Model Foundation; Luxembourg Ministry of Development Cooperation and Humanitarian Aid; Luxembourg Ministry of Health; Médecins Sans Frontières; Swiss Federal Office of Public Health; UK Foreign, Commonwealth & Development Office (previously the UK Department for International Development). GARDP were consulted on study design. Members of the funder participated as authors on the study. All authors had the opportunity to access and verify the data, and all authors were responsible for the decision to submit the manuscript for publication.

**Competing interests:** I have read the journal's policy and the authors of this manuscript have the following competing interests: Members of the funder (MB, NK, SE) participated as authors on the study, reviewed the manuscript, and approved the final manuscript as submitted. All other authors have declared that no competing interests exist.

Abbreviations: AMR, antimicrobial resistance; AUROC, area under the receiver operator curve; CLSI, Clinical and Laboratory Standard Institute; CRO, carbapenem-resistant organism; CSF, cerebrospinal fluid; EMLc, Essential Medicine List for Children; EOA, external quality assurance; ESBL, extended-spectrum beta-lactamase; EUCAST, European Committee on Antimicrobial Susceptibility Testing; FP, fractional polynomial; HAI, healthcare-associated infection; LMIC, lowand middle-income country; MDR, multidrug resistant; pSBI, possible serious bacterial infection; SMART, Sequential Multiple Assessment Randomized Trial; SOC, standard of care; WHO, World Health Organization. Shenzhen, China, 25 Laboratory of Medical Microbiology, University of Antwerp, Antwerp, Belgium, 26 Neonatal Unit, Department of Pediatrics, Queen Sirikit National Institute of Child Health, College of Medicine, Rangsit University, Bangkok, Thailand, 27 Neonatology Dept, School of Medicine, Faculty of Health Sciences, Aristotle University and Hippokration General Hospital, Thessaloniki, Greece, 28 Department of Global Public Health, Karolinska Institutet, Stockholm, Sweden, 29 Department of Paediatrics and Child Health, College of Health Sciences, Makerere University and MUJHU Care, Kampala, Uganda, 30 Makerere University - Johns Hopkins University Research Collaboration, Kampala, Uganda, 31 Amsterdam UMC, University of Amsterdam, Emma Children's Hospital, Department of Global Health, Amsterdam, the Netherlands, 32 Chiangrai Prachanukroh Hospital, Chiang Rai, Thailand, 33 World Health Organization, Maternal, Newborn, Child and Adolescent Health Department, Geneva, Switzerland, 34 Department of Infectious Diseases, Shenzhen Children's Hospital, Shenzhen, China, 35 Department of Paediatrics and Child Health, Charlotte Maxeke Johannesburg Academic Hospital, Johannesburg, South Africa, 36 Infectious Diseases Unit, 3rd Dept Pediatrics, School of Medicine, Faculty of Health Sciences, Aristotle University and Hippokration General Hospital, Thessaloniki, Greece, 37 School of Clinical Medicine, Faculty of Health Sciences, University of the Witwatersrand, Johannesburg, South Africa, 38 Department of Neonatology, Beijing Children's Hospital, Capital Medical University, National Centre for Children's Health, Beijing, China, 39 Department of Neonatology, Children's Hospital, Capital Institute of Pediatrics, Yabao Road, Chaoyang District, Beijing, China

<sup>‡</sup> These authors share first authorship on this work. \* <u>nrussell@sgul.ac.uk</u>

# Abstract

# Background

There is limited data on antibiotic treatment in hospitalized neonates in low- and middleincome countries (LMICs). We aimed to describe patterns of antibiotic use, pathogens, and clinical outcomes, and to develop a severity score predicting mortality in neonatal sepsis to inform future clinical trial design.

# Methods and findings

Hospitalized infants <60 days with clinical sepsis were enrolled during 2018 to 2020 by 19 sites in 11 countries (mainly Asia and Africa). Prospective daily observational data was collected on clinical signs, supportive care, antibiotic treatment, microbiology, and 28-day mortality. Two prediction models were developed for (1) 28-day mortality from baseline variables (baseline NeoSep Severity Score); and (2) daily risk of death on IV antibiotics from daily updated assessments (NeoSep Recovery Score). Multivariable Cox regression models included a randomly selected 85% of infants, with 15% for validation.

A total of 3,204 infants were enrolled, with median birth weight of 2,500 g (IQR 1,400 to 3,000) and postnatal age of 5 days (IQR 1 to 15). 206 different empiric antibiotic combinations were started in 3,141 infants, which were structured into 5 groups based on the World Health Organization (WHO) AWaRe classification. Approximately 25.9% (n = 814) of infants started WHO first line regimens (Group 1—Access) and 13.8% (n = 432) started WHO second-line cephalosporins (cefotaxime/ceftriaxone) (Group 2—"Low" Watch). The largest group (34.0%, n = 1,068) started a regimen providing partial extended-spectrum beta-lacta-mase (ESBL)/pseudomonal coverage (piperacillin-tazobactam, ceftazidime, or fluoroquino-lone-based) (Group 3—"Medium" Watch), 18.0% (n = 566) started a carbapenem (Group 4 —"High" Watch), and 1.8% (n = 57) a Reserve antibiotic (Group 5, largely colistin-based),

and 728/2,880 (25.3%) of initial regimens in Groups 1 to 4 were escalated, mainly to carbapenems, usually for clinical deterioration (n = 480; 65.9%).

A total of 564/3,195 infants (17.7%) were blood culture pathogen positive, of whom 62.9% (n = 355) had a gram-negative organism, predominantly *Klebsiella pneumoniae* (n = 132) or *Acinetobacter* spp. (n = 72). Both were commonly resistant to WHO-recommended regimens and to carbapenems in 43 (32.6%) and 50 (71.4%) of cases, respectively. MRSA accounted for 33 (61.1%) of 54 *Staphylococcus aureus* isolates.

Overall, 350/3,204 infants died (11.3%; 95% Cl 10.2% to 12.5%), 17.7% if blood cultures were positive for pathogens (95% Cl 14.7% to 21.1%, n = 99/564). A baseline NeoSep Severity Score had a C-index of 0.76 (0.69 to 0.82) in the validation sample, with mortality of 1.6% (3/189; 95% Cl: 0.5% to 4.6%), 11.0% (27/245; 7.7% to 15.6%), and 27.3% (12/44; 16.3% to 41.8%) in low (score 0 to 4), medium (5 to 8), and high (9 to 16) risk groups, respectively, with similar performance across subgroups. A related NeoSep Recovery Score had an area under the receiver operating curve for predicting death the next day between 0.8 and 0.9 over the first week. There was significant variation in outcomes between sites and external validation would strengthen score applicability.

# Conclusion

Antibiotic regimens used in neonatal sepsis commonly diverge from WHO guidelines, and trials of novel empiric regimens are urgently needed in the context of increasing antimicrobial resistance (AMR). The baseline NeoSep Severity Score identifies high mortality risk criteria for trial entry, while the NeoSep Recovery Score can help guide decisions on regimen change. NeoOBS data informed the NeoSep1 antibiotic trial (ISRCTN48721236), which aims to identify novel first- and second-line empiric antibiotic regimens for neonatal sepsis.

# **Trial registration**

ClinicalTrials.gov, (NCT03721302).

# Author summary

# Why was this study done?

- Increasing trends in antimicrobial resistance (AMR) disproportionately affect neonates and young infants with sepsis in LMIC settings and undermine the effectiveness of WHO-recommended antibiotics.
- Despite this, longitudinal data on antibiotic management strategies and outcomes of hospitalized neonates and young infants with sepsis in low- and middle-income country (LMIC) settings are extremely limited, impeding the design of robust antibiotic trials.
- There is limited data to define risk stratification, inclusion, and escalation criteria in trials of sepsis in hospitalized neonates and young infants.

### What did the researchers do and find?

- In this large global, prospective, hospital-based observational study across 4 continents, including LMIC settings, there was a high mortality among infants with culture positive sepsis (almost 1 in 5), and a significant burden of antibiotic resistance.
- This study highlights wide variations in standard of care (SOC) for sepsis in neonates and young infants, with more than 200 different antibiotic combinations, significant divergence from WHO-recommended regimens, and frequent switching of antibiotics.
- A NeoSep Severity Score that defined patterns of mortality risk at baseline was developed from 4 non-modifiable and 6 modifiable factors that are feasible to measure across a range of LMIC hospital contexts.
- A NeoSep Recovery Score including the same modifiable factors (with the addition of cyanosis) predicted mortality on the following day during treatment.

### What do these findings mean?

- These data demonstrate that patterns of routine antibiotic use are now markedly divergent from global guidance.
- There is an urgent need for large pragmatic randomized controlled trials to address optimal empiric first- and second-line antibiotic treatment strategies in LMIC hospital settings with a significant AMR burden.
- The wide range of multiple antibiotic regimens routinely used as SOC suggests the need for novel trial designs.
- The NeoSep Severity Score and NeoSep Recovery Score informed inclusion and escalation criteria in the NeoSep1 antibiotic trial (ISRCTN48721236) that aims to identify novel first- and second-line empiric antibiotic regimens for neonatal sepsis.

# Introduction

Sepsis is responsible for a significant burden of disease in neonates and young infants, both as a primary cause of death and as a frequent contributor [1,2]. Access to facility-based delivery and care, including antibiotics, has not reduced mortality sufficiently to achieve the Sustainable Development Goal targets in many low- and middle-income countries (LMICs) [3]. Antimicrobial resistance (AMR) increasingly threatens to undermine the effectiveness of antibiotics and potentially slow progress in reducing mortality, particularly in LMICs [4–10], with AMR-attributable neonatal deaths recently estimated between 140,000 [10] and 214,000 annually [11].

Recent large-scale antibiotic trials in neonates and young infants in LMIC settings have largely focused on simplification of first-line antibiotic regimens with oral amoxicillin and short course gentamicin [12,13]. These have been based in primary healthcare settings and included populations with case fatality below 2%. An increasing global proportion of newborns are delivered in facilities [14], where sepsis case fatality rates and the burden of AMR are

greater. Despite this, there is limited high-quality evidence generated from LMIC neonatal inpatient settings to guide empiric antibiotic treatment [4,15]. Published observational data largely involve single-center studies reporting non-systematically collected microbiological data, which are rarely accompanied by detailed clinical and antibiotic use data. Data regarding global antibiotic use often rely on point prevalence surveys, with limited information on patterns of switching and duration [16].

NeoOBS is a prospective multicountry observational study in which we collected detailed daily longitudinal data on clinical features, microbiology, antibiotic use, switching, and outcomes of neonatal sepsis in hospital settings, predominantly in LMICs. The objectives were to describe variation in clinical presentation and patterns of hospital-based antibiotic use and to develop 2 linked clinically based scores adapted from the World Health Organization (WHO) possible serious bacterial infection (pSBI) criteria, relevant to hospitalized neonates with sepsis in LMICs: (1) a sepsis severity score to predict 28-day mortality from factors known at sepsis presentation; and (2) a recovery score to predict the daily risk of death on treatment with IV antibiotics using daily updated assessments of clinical status. The aim of the study was to inform inclusion criteria, empiric and second-line treatment, and criteria for switching antibiotics in the design of hospital-based neonatal antibiotic trials [17].

# Methods

### Study design and participants

Hospitalized infants <60 days of age with a new episode of clinically suspected sepsis were enrolled between 2018 and 2020 in 19 hospital sites across 11 countries in Asia (Bangladesh, China, India, Thailand, and Vietnam), Africa (Kenya, South Africa, and Uganda), Europe (Italy and Greece), and South America (Brazil). Sites were selected after conducting a feasibility study [18] to represent diverse regions and to include secondary and tertiary referral hospitals, public facilities, and facilities with varying proportions of in-born and out-born infants, with access to microbiology.

Infants were eligible if the local physician had decided to treat the infant with antibiotics for a new episode of sepsis meeting the inclusion criteria (S1 Fig), derived by combining clinical and laboratory criteria from WHO pSBI [19] and EMA Criteria for neonatal sepsis trials [20]. To allow for variation in access to laboratory testing, and ensure generalizability to varying LMIC hospital contexts, laboratory values were not mandatory. A minimum of 2 clinical, or 1 clinical and 1 laboratory sepsis criteria, were required for inclusion, and up to 200 infants per site were enrolled according to a sampling frame adapted to local case volume and activity (see S1 Appendix). Infants were excluded if an alternative primary diagnosis other than sepsis was suspected, or a serious non-infective comorbidity was expected to cause death within 72 h. Previous antibiotic use was not an exclusion criterion as long as a new antibiotic regimen was being started after a blood culture for a distinct new episode of sepsis. Sepsis episodes occurring >48 h after admission, defined by time of blood culture, were defined as healthcare-associated infections (HAIs).

Ethical approval was obtained from St. George's, University of London (SGUL) Research Ethics Committee and sites' local, central or national ethics committees and other relevant local bodies, where required. This study is reported using (1) the Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) guideline (S1 STROBE Checklist), with its design guided by the STROBE-NI framework [21]; and (2) the Transparent Reporting of a multivariable prediction model for Individual Prognosis Or Diagnosis (TRIPOD) guidelines [22]. The study was registered with ClinicalTrials.gov (NCT03721302).

# Procedures

After written consent from parents, baseline demographic and clinical data were collected, followed by prospective daily collection of observational data including multiple clinical parameters, laboratory investigations, and microbiological results. Antibiotic data were collected daily including drug, dose, route, duration, switching, and reasons given for any changes. Clinical data collection was required up to 48 h after the completion of antibiotic therapy or discharge if sooner. Aside from a mandatory blood culture at enrolment and daily monitoring of vital signs, all clinical observations and investigations were performed according to routine local site practices. Infants were followed until 28 days after enrolment in-person if still hospitalized or by telephone post-discharge. A final diagnosis was documented by clinicians, as were primary and secondary causes of death, and any clinical illness or readmission occurring after discharge and within 28 days of enrolment.

Data were collected by research and clinical staff based on clinical observation and routine source documentation (e.g., medical and nursing notes, vital signs, and prescription charts), and entered and managed using REDCap electronic data capture tools [23] hosted at SGUL (details on data monitoring in S1 Appendix).

### Microbiology/laboratory assessments

Laboratory analysis was performed in each site following local practice, with standard operating procedures developed to optimize procedures including blood culture technique, and antibiotic susceptibility testing. A locally defined algorithm was used to classify contaminants and pathogens by site clinicians and microbiologists. External validation of the capability of laboratories to detect multidrug resistant (MDR) gram-negative pathogens from each site was evaluated objectively by testing an external quality assurance (EQA) panel sent from the central laboratory at Laboratory of Medical Microbiology (LMM) at the University of Antwerp (UA).

European Committee on Antimicrobial Susceptibility Testing (EUCAST) or Clinical and Laboratory Standard Institute (CLSI) guidelines and interpretive algorithms were used to interpret reported antibiotic susceptibility testing data for all organisms. EUCAST 2019 breakpoints table (v9.0) and guidelines were used as this was when most data was reported. In most cases, where susceptibility to a particular antibiotic was reported, this was used to determine susceptibility. If the organism was considered intrinsically resistant to a particular antibiotic according to EUCAST, then it was coded as resistant regardless of reported susceptibility. If susceptibility to a particular antibiotic was not reported, susceptibility results of a different antibiotic in the same class were used to determine susceptibility (e.g., susceptibility of organism to another aminoglycoside if gentamicin susceptibility was not reported); if no other antibiotic in that class had reported susceptibility, then susceptibility was coded as unknown.

### Analysis of antibiotic patterns of use

The initial antibiotic regimen was defined as the first new antibiotic(s) started within 24 h from baseline blood culture (including 3 h pre baseline culture). To structure the analysis and reporting of the multiple regimens used, a novel method of grouping antibiotics was derived, based on the Essential Medicine List for Children (EMLc) AWaRe classification (Access, Watch, Reserve) [24], with the "Watch" category divided into 3 distinct groups of "Low/ Medium/High Watch" based on inclusion in current WHO guidelines (Low Watch) and likelihood of resistance generation with regimens outside WHO recommendations (Medium or High Watch) [25]. Antibiotic groups were defined by the main "stem" in the antibiotic combination: Group 1 antibiotics included a first-line WHO-recommended penicillin-based regimen (e.g., ampicillin and gentamicin) (Access), Group 2 included third-generation

cephalosporin (e.g., cefotaxime/ceftriaxone)-based WHO regimens ("Low" Watch), Group 3 included regimens with partial anti-extended-spectrum beta-lactamase (ESBL) or pseudomonal activity (e.g., piperacillin-tazobactam/ceftazidime/fluoroquinolone-based) ("Medium" Watch), and Group 4 included carbapenems (e.g., meropenem) ("High" Watch). Group 5 antibiotics included Reserve antibiotics targeting carbapenem-resistant organisms (CROs) (e.g., colistin). Aminoglycosides (e.g., gentamicin/amikacin), glycopeptides (e.g., vancomycin/teicoplanin), and metronidazole used in combination regimens were classified as additional coverage and did not define the main antibiotic "stem" for the grouping. All antifungals and antivirals were excluded from the antimicrobial treatment data as these were not relevant to the analysis. Escalation of treatment was defined as a switch to a higher group antibiotic, and de-escalation was defined as switching to a lower group or discontinuation of the "stem" antibiotic while continuing with an additional coverage antibiotic.

### Statistical analysis

Baseline characteristics are presented as raw numbers, without any weighting (S1 Appendix). The prespecified primary outcome was mortality through 28 days post-enrolment, analyzed using Kaplan–Meier and Cox proportional hazards regression with site-level random effects. Time was measured from the initial blood culture sample, censoring at the earliest of day 28, withdrawal or last contact if lost post-discharge. Aligned with the WHO pSBI criteria, we developed 2 prediction models and corresponding risk scores: (1) a baseline NeoSep Severity Score to predict 28-day mortality from factors known at sepsis presentation; and (2) a NeoSep Recovery Score to predict the daily risk of death while treated with IV antibiotics from daily updated assessments of clinical status.

**NeoSep Severity Score.** We constructed a Cox regression model for 28-day mortality based on baseline clinical parameters known at sepsis presentation (i.e., before availability of microbiological results) and used this to develop a neonatal sepsis severity score. Baseline was defined as up to 24 h from the time blood cultures were taken. Candidate predictors selected for initial consideration had missing values in <10% of the included infants and had been found to be predictive in other studies [26] or were a priori determined to be clinically important and not highly correlated with other factors (S1 Table). Factors that are not usually available in low-income settings including laboratory results (missing in >10%), and clinical signs with prevalence <5% were not considered. All analyses were based on available data, and imputation methods were not used. A 15% randomly selected sample per site was reserved for model validation and not used in any model development. This validation procedure was predefined because of expected large differences between sites in mortality, instead of randomly choosing 1 or 2 sites for validation which then might have turned out to be unrepresentative. In the remaining 85% of infants, model development used backwards elimination (exit p = 0.05) to identify independent predictors of mortality, considering both categorical (e.g., presence of sepsis signs) and continuous predictors (birth weight, gestational age, time in hospital, age at baseline, oxygen saturation, respiratory rate, heart rate, and temperature), taking into account the potential for nonlinear relationships (e.g., where both high and low values are associated with poor outcomes) by using fractional polynomials (FPs). FPs were implemented using the mfp command in Stata with power (-2, -1, -0.5, 0, 0.5, 1, 2) and significance level of 0.05 for testing between FP models of different degrees. Initial variable selection was done on complete cases that comprised 2,313/2,726 (85%) infants in the training set; final models were re-fitted to complete cases for the included factors (2,705/2,726 (99%) infants in the derivation sample). Interactions between birth weight and clinical predictors were examined and 1 strong interaction (p < 0.001) was found with ventilation support; however, this was not included in

the final model in favor of simplicity for use of the prediction model in clinical practice and because including additional points for the interaction made very little difference to the C-statistic. The proportional hazards assumption was checked based on Schoenfeld residuals, and no deviations were found (p > 0.05). A points-based risk score, where each predictor of death is assigned a number of points, was then developed from model coefficients: For each of the categorical factors, coefficients were divided by the smallest of the coefficients (lethargy) and rounded to the nearest integer. For the continuous factors, a clinically relevant reference value was chosen (birth weight: 3,000 g; temperature: 37°C; gestational age: 39 weeks; time in hospital: 14 days), and the number of points associated with lower/higher values was then based on the difference to the reference. This initial score was then further simplified to provide a more feasible and pragmatic scale for use in LMIC by assigning 1 score point for a predictor if its value in the initial score was <3, 2 score points if its initial value was 3 to 5 points, and 3 score points if its initial value was >6 points. For each participant, points were added with a higher score indicating a higher risk of death. Harrell's C-index with bootstrapped confidence intervals was used as a measure of discrimination of the prognostic Cox models, and the Hosmer-Lemeshow goodness of fit test was applied after running a logistic model with death as the independent and the score as the dependent variable as a measure of calibration.

We compared the NeoSep Severity Score with a score based on WHO pSBI [19], allocating 1 point for each of the following 6 signs: fever ( $\geq$ 38°C), low body temperature (<35°C), movement only on stimulation, feeding poorly, fast breathing ( $\geq$ 60 breaths per minute on days 0 to 6), and severe chest in-drawing. Whereas the first 5 signs were directly reported/measured in NeoOBS, the latter was only reported as part of a composite respiratory sign (severe chest wall in-drawing, increased oxygen requirement, or need for ventilation); for the score calculation, 1 point was given if this was reported.

**NeoSep Recovery Score.** To estimate the association between daily risk of death after initiating IV antibiotics and time-updated factors, and hence develop a recovery score, we used Cox proportional hazards regression with time-varying independent factors. We used the same samples as described above for model derivation and validation. Infants were censored after stop of all IV antibiotics (cause-specific model). We started model building with all clinical predictors included in the baseline severity score as time-updated factors; unmodifiable infant and birth characteristics were excluded because they cannot evolve and would have restricted interpretation of the recovery score (e.g., preterm babies would have a minimum score of 1 and could have never reached "full recovery" with a score of 0). Forward selection (entry p = 0.05) was then used to identify any additional independent time-updated clinical predictors. FPs were used for continuous factors as described above. For vital parameters, the last value was used per assessment period. To avoid selecting factors representing the mechanism of dying rather than being predictors of subsequent death, clinical parameters reported on the previous day were used as predictors of death on the current day in all time-updated models. A points-based risk score was then derived similarly to the baseline severity score as described above. Discrimination was assessed using time-updated area under the receiver operator curves (AUROCs), ignoring censoring as this was low (<2% overall). To examine the potential usefulness of the recovery score for informing the decision to potentially switch to second-line antibiotics in future empiric clinical trials, we focused on day 2 (48 to 72 h) post baseline, a key decisional time point during clinical management when culture results become available and response to treatment is commonly evaluated.

Analysis of antibiotic use. Cumulative incidence of antibiotic escalation or stop of all IV antibiotics was estimated with death as competing risk. Multivariable logistic regression with backwards elimination (exit p = 0.05) adjusted for site was used to analyze factors associated with starting groups 3 to 5 versus groups 1 to 2 regimens in sites with at least 10% of infants in

either. Candidate factors were known at sepsis presentation and included birth weight, gestational age, postnatal age, time in hospital, central line or indwelling catheter, intravenous (IV) antimicrobials in previous 24 h, signs of meningitis, previous positive culture, previous surgery, and the NeoSep Recovery Score. FPs were used to take into account the potential for nonlinear relationships. To analyze whether time from baseline culture to the start of new antibiotic regimen or type of pathogen were associated with mortality, respectively, we adjusted for the NeoSep Severity Score. Analyses used Stata version 16.1.

# Results

A total of 3,204 infants (90.4% neonates aged <28 days, n = 2,895; 42.1% female, n = 1,348) were recruited from 20 August 2018 to 29 February 2020 (Table 1 and S2 Fig). Sites included varying populations of infants and levels of care (S3–S6 Figs and S2 Table). The median postnatal age was 5 (IQR 1 to 15) days, and 3,088 (96.4%) infants had been born in a hospital/facility (1,550 in the enrolling facility), 1,412 (44.3%) by cesarean section (969 as an emergency), and 71 (2.2%) had previously been treated for an episode of culture-positive sepsis. The median (IQR) gestational age at birth was 37 (31 to 39) weeks, with birth weight 2,500 g (1,400 to 3,000 g). At enrollment, 69.1% (n = 2,215) infants had been hospitalized since birth, and 30.9% (n = 989) were admitted from the community. A total of 2,759 (86.1%) were recruited in a neonatal unit. Among 309 (9.6%) infants enrolled aged ≥28 days, the majority (n = 181; 58.6%) were either ex-premature (n = 146; 47.4%) and/or had been admitted during the neonatal period (n = 136, 44.2%).

The most common previously identified risk factors for sepsis other than prematurity were preterm premature rupture of membranes (14.5%, n = 466), prolonged rupture of membranes (>18 h) (10.2%, n = 328), pre-labor rupture of membranes at term (9.4%, n = 300), presence of an indwelling central vascular catheter (8.2%, n = 262), intrapartum fever >38°C (3.5%, n = 112), chorioamnionitis (2.8%, n = 75), known maternal Group B streptococcus (GBS) colonization (1.4%, n = 46) and previous surgery (abdominal or for congenital malformations: 3.0%, n = 95). Of note, 8.3% (n = 265) had a congenital anomaly (S3 Table), and 7% (n = 220) of infants were exposed to maternal HIV. A total of 1,318 (41.1%) sepsis episodes were classified as healthcare-associated (occurring >48 h after hospital admission).

# Clinical and laboratory findings

A median of 4 (IQR 2 to 5) clinical signs were present at baseline, the most common being respiratory (65.8%, n = 2,107), difficulty feeding (45.7%, n = 1,464), lethargy or reduced movement (35.3%, n = 1,131), abdominal distension (24.3%, n = 777), and evidence of shock (21.3%, n = 683), with prevalence decreasing over time (Table 1 and S7 Fig). Signs suggestive of meningitis were reported in <10% of infants (irritability, convulsions, abnormal posturing, bulging fontanelle) (S8 Fig). The availability of laboratory values varied by site. At baseline, a base excess <-10 mmol/L was documented in 352 of 1,492 with results available (23.6%), lactate >2 mmol in 1,034/1,283 (80.6%), raised CRP >10 mg/L in 1,306/2,286 (57.1%), abnormal white blood cell count (<4 or >20 × 10<sup>9</sup> cells/L) in 875/2,800 (31.3%), and thrombocytopenia (<150 × 10<sup>9</sup> cells/L) in 619/2,776 (22.3%) (S4 Table).

# Patterns of empiric antibiotic use

A total of 1,180 (36.8%) infants had a history of previous intravenous antibiotic exposure, and 742 (23.2%) had been receiving an intravenous antibiotic in the previous 24 h before starting new antibiotics for the new episode of sepsis (85.3% treatment (n = 633) and 14.2% prophylaxis (n = 105)). Median time from baseline culture being taken to new intravenous antibiotic

### Table 1. Baseline characteristics.

Infant characteristics			N = 3,204			
	WHO region of enrolment:	Africa	998 (31.1%)			
		Americas	79 (2.5%)			
		Southeast Asia	1,201 (37.5%)			
		Europe	121 (3.8%)			
		Western Pacific	805 (25.1%)			
	Age at baseline (days), median (IQ	Age at baseline (days), median (IQR)				
	Sex	Male	1,854 (57.9%)			
		Female	1,348 (42.1%)			
		Indeterminate/intersex	2 (0.1%)			
	Birth weight (grams), median (IQ	Birth weight (grams), median (IQR)				
	Gestational age at birth (weeks), m	37 (31, 39)				
	Estimated gestational age at birth,					
		Extremely preterm (<28 weeks)	227 (7.1%)			
		Very preterm (28 to <32 weeks)	607 (19.0%)			
		Moderately preterm (32 to <37 weeks)	694 (21.7%)			
		Term (≥37 weeks)	1,674 (52.3%)			
Birth history	Birth status	Hospitalized since birth	2,215 (69.1%)			
		Admitted from home/community	989 (30.9%)			
	Time from admission to enrolmer	Time from admission to enrolment (hours), median (IQR)				
	Mode of delivery:	Vaginal	1,774 (55.4%)			
		Emergency cesarean section	969 (30.2%)			
		Elective cesarean section	443 (13.8%)			
	Congenital anomalies	265 (8.3%)				
Common subgroups	Early-onset (age <48 h)	1,066 (39.3%)				
	Late onset community presenting,	Late onset community presenting, term				
	Late onset healthcare associated, p	936 (34.5%)				
	Other	Other				
Antibiotics	On IV antibiotics in previous 24 h	On IV antibiotics in previous 24 h				
Previous surgery	Abdominal surgery or for congeni	Abdominal surgery or for congenital malformations				
Supportive care at baseline	IV fluid (supportive/feeding)	2,497 (77.9%)				
	Thermal care	Incubator care	984 (30.7%)			
		Heated mattress	115 (3.6%)			
		Overhead heater	(33.7%)			
		Kangaroo mother care	163 (5.1%)			
	Oxygen supplementation	1,894 (59.1%)				
	Ventilation support:	Non-invasive ventilation	719 (22.4%)			
		Invasive ventilation	701 (21.9%)			
	Phototherapy	522 (16.3%)				
	Transfusion red blood cells	221 (6.9%)				
	Transfusion platelets	64 (2.0%)				
	Fresh frozen plasma	118 (3.7%)				
	Feeding:	Breast/formula/other milk	1,953 (61.0%)			
		TPN	527 (16.4%)			
		Nasogastric tube	1,432 (44.7%)			
Baseline culture	Negative	Negative				
	Contaminant (presumed non-path	Contaminant (presumed non-pathogen)				
	Pathogen	Pathogen				
	Indeterminate	12 (0.4%)				

#### Table 1. (Continued)

Pathogens	Klebsiella pneumoniae	132 (4.1%)			
	Coagulase-negative Staphylococci	84 (2.6%)			
	Acinetobacter spp. *	72 (2.3%)			
	Staphylococcus aureus	54 (1.7%)			
	Escherichia coli	47 (1.5%)			
	Enterobacter spp.	39 (1.2%)			
	Serratia spp.	20 (0.6%)			
	Streptococcus agalactiae	19 (0.6%)			
	Other gram-negative bacteria **	65 (2.0%)			
	Other gram-positive bacteria ***	42 (1.3%)			
	Fungi ****	21 (0.7%)			
Vital parameters	Oxygen saturation (%), median (IQ	96 (93, 98)			
-	Respiratory rate, median (IQR)	52 (44, 60)			
	Heart rate, median (IQR)	Heart rate, median (IOR)			
	Temperature (°C)	<35.5	58 (1.8%)		
		35.5–37.9	2,736 (85.6%)		
		≥38-<39	320 (10.0%)		
		≥39	84 (2.6%)		
Clinical signs at baseline (≥5%)	Respiratory signs <sup>‡</sup>	2,107 (65.8%)			
	Difficulty feeding T	1,465 (45.7%)			
	Lethargy/reduced movement	Neither	2,073 (64.7%)		
		Lethargy only	803 (25.1%)		
		Reduced/no movement	328 (10.2%)		
	Abdominal distension	777 (24.3%)			
	Evidence of shock	683 (21.3%)			
	Apnoea	550 (17.2%)			
	Jaundice requiring phototherapy	494 (15.4%)			
	Grunting	435 (13.6%)			
	Hypotonia/floppiness	435 (13.6%)			
	Cyanosis	384 (12.0%)			
	Irritability	322 (10.0%)			
	Vomiting	287 (9.0%)			
	Convulsions	236 (7.4%)			

\**Acinetobacter baumannii*: n = 64; *Acinetobacter iwoffii*: n = 2; unspecified: n = 6.

\*\* Enterococcus faecalis: n = 14; Enterococcus faecium: n = 10; Bacillus spp.: n = 4; Streptococcus pneumoniae: n = 3; Streptococcus pyogenes: n = 3; Aerococcus viridans: n = 1; Corynebacterium spp.: n = 1; Dermabacter hominis: n = 1; Listeria monocytogenes: n = 1; Streptococcus gallolyticus: n = 1; Streptococcus salivarius: n = 1; Streptococcus viridans: n = 1; Strept

\*\*\**Elizabethkingia meningoseptica*: n = 15; Burkholderia spp.: n = 12; Citrobacter spp.: n = 10; Elizabethkingia anophelis: n = 7; Klebsiella oxytoca: n = 7; Pseudomonas aeruginosa: n = 5; Proteus mirabilis: n = 2; Campylobacter coli: n = 1; Klebsiella spp.: n = 1; Morganella morgannii: n = 1; Pseudomonas putida: n = 1; Pseudomonas stutzeri: n = 1; Salmonella spp.: n = 1; Sphingomonas paucimobilis: n = 1; Stenotrophomonas maltophilia: n = 1.

\*\*\*\*Candida albicans: n = 13; Candida parapsilosis: n = 2; Candida pelliculosa: n = 2; Candida auris: n = 1; Candida tropicalis: n = 1; Kodamaea ohmeri: n = 1; Wickerhamomyces anomalus: n = 1.

\*Severe chest wall in-drawing, increased requirement for oxygen, or respiratory support.

<sup>T</sup>Observed or reported, including feeding intolerance.

IV, intravenous.

https://doi.org/10.1371/journal.pmed.1004179.t001







**Fig 1. Antibiotic treatment.** (A) Initial empiric baseline therapy overall and by site. (B) First change of initial regimen, by baseline regimen. (C) Cumulative incidence of stopping all IV antibiotics (death on IV antibiotics as competing risk). Group 1 = First-line WHO-recommended penicillinbased regimen (e.g., ampicillin and gentamicin) (Access). Group 2 = third-generation cephalosporin (e.g., cefotaxime/ceftriaxone)-based WHO regimens ("Low" Watch). Group 3 = regimens with partial anti-ESBL or pseudomonal activity (e.g., piperacillin-tazobactam/ceftazidime/ fluoroquinolone-based) ("Medium" Watch). Group 4 = Carbapenems ("High" Watch). Group 5 = Reserve antibiotics targeting carbapenem-resistant organisms (e.g., colistin). ESBL, extended-spectrum beta-lactamase; IV, intravenous.

treatment being started for the distinct sepsis episode was 1 h (IQR 0 to 3); 2,913 (90.9%) started within 8 h, and 228 (7.1%) within 8 to 24 h (S9 Fig); 63 (2.0%) infants did not start any new antibiotic in the first 24 h.

There were 206 different combinations of empiric antibiotics started at baseline in this hospital-based cohort. The most frequent regimens are reported in S5 Table, and these were grouped as described in the methods. Of 3,141 infants who started new antibiotics within 24 h, 25.9% (n = 814) were started on a WHO Neonatal Sepsis Guideline recommended first-line penicillin-based regimen (Group 1—Access), and 13.8% (n = 432) were started on a WHO second-line cefotaxime or ceftriaxone-based combination (Group 2—"Low" Watch) (Fig 1A). The largest group (34.0%, n = 1,068) were started on a regimen providing partial ESBL/pseudomonal coverage (piperacillin-tazobactam, ceftazidime, or fluoroquinolone-based) (Group 3—"Medium" Watch). Of Group 3 regimens, ceftazidime ± amikacin (n = 436; 13.9%) and piperacillin/tazobactam ± amikacin (n = 410; 13.1%) were most common. Approximately

18.0% (n = 566) of initial regimens were carbapenem-based (Group 4—"High" Watch), among whom meropenem ± vancomycin (n = 447; 14.2%) was the most common combination; 1.8% (n = 57) of initial regimens were classified as Group 5 antibiotics targeting CROs, predominantly colistin-based (Group 5—Reserve) (S10–S14 Figs). An "Other" group (n = 204) consisted of more rarely used local regimens not on the WHO EMLc, or regimens which did not include a new antibiotic "stem" that was used to define Groups 1 to 5 (e.g., aminoglycoside or glycopeptide given alone or in combination with each other) (S15 Fig). Cefoperazone/sulbactam (n = 99) was the most common antibiotic in this category, but was used as initial regimen only in India (n = 86; 14.5%), China (n = 11; 1.9%), and Vietnam (n = 2; 1.0%).

There was wide variation between sites in the frequency of empiric use of different antibiotic groups (Fig 1A). Some sites used predominantly Group 1 antibiotics as the initial regimen, and others often started immediately with Group 3 or 4 regimens, or a mixture of all groups. The most frequently prescribed empiric regimens used for HAIs were meropenem  $\pm$  vancomycin (28.2%, 361/1,280), followed by piperacillin/ tazobactam  $\pm$  amikacin (17.5%, n = 224), ceftazidime  $\pm$  amikacin (6.7%, n = 86), colistin (4.3%, n = 55), and cefoperazone/subactam (3.9%, n = 50). Ampicillin + gentamicin was the most common regimen for non-HAI (19.7%, 366/1,861), followed by ceftazidime  $\pm$  amikacin (18.8%, n = 350) and piperacillin/tazobactam  $\pm$  amikacin (10.0%, n = 186) (S6 Table).

Adjusting for site, predictors of starting empiric therapy with Groups 3 to 5 rather than Groups 1 to 2 antibiotics included lower birth weight (OR = 0.57 per additional kg, 95% CI 0.47 to 0.69), presence of a central vascular catheter (OR = 3.48, 95% CI 1.74 to 6.94), previous antibiotics at enrollment (OR = 5.71, 95% CI 3.73 to 8.77), previous culture positive sepsis (OR = 25.71, 95% CI 3.00 to 220.7), longer time in hospital (48 versus 0 hours: OR = 4.41, 95% CI 3.41 to 5.69), previous surgery (OR = 5.18, 95% CI 1.65 to 16.28), and higher sepsis severity (OR = 1.27 per additional score point, 95% CI 1.16 to 1.40) (see S7 Table).

### Antibiotic switching

After initial therapy, 728/2,880 (25.3%) who started on Group 1 to 4 regimens were escalated to a higher group regimen, the majority switching within the first days of treatment (S16–S18 Figs), and 258/814 (31.7%) infants escalated from Group 1 (ampicillin/gentamicin) regimens, the majority of whom switched to Group 2 (cefotaxime/ceftriaxone-based) regimens (61.6%, n = 159) (Fig 1B). A total of 101/432 (23.4%) infants escalated from Group 2 regimens, mostly directly to a Group 4 carbapenem-based regimen (62.4%, n = 63) rather than a Group 3 partial ESBL/pseudomonal activity regimen (32.7%, n = 33); 287/1,068 (26.9%) escalated from Group 3 regimens, and of 566 infants starting treatment with carbapenems (Group 4), 82 (14.5%) escalated therapy to a colistin-containing regimen. Common reported reasons for first escalation of antibiotics overall included clinical deterioration (65.9%, n = 480), microbiology results (15.4%, n = 112: n = 48 pathogen identification, n = 37 susceptibility, n = 27 gram stain), and worsening inflammatory biomarkers (9.8%, n = 71). De-escalation of antibiotics was rare (173/2,937; 5.9%).

Cumulative incidence of stopping intravenous treatment within 7 days from baseline culture was 38.9% (95% CI 37.2% to 40.6%) overall (Fig 1C); 45.8% (95% CI 43.8% to 47.6%) in pathogen-negative and 6.9% (95% CI 5.0% to 9.2%) in pathogen-positive cases. After stopping intravenous antibiotics, 350/2,803 (12.5%) switched to oral therapy, and 289 of 2,803 infants who had stopped (10.3%), restarted intravenous antibiotics during the original hospital stay, and a further 84 infants after discharge. A total of 115 (3.4%) were still on uninterrupted antibiotic treatment at day 28. Median total number of days on intravenous antibiotics during the 28 days follow-up was 8 (IQR 6 to 14) days (S19 Fig). Intramuscular use of antibiotics was only reported for 2 infants (<0.1%).

# Microbiology

Initial blood culture results were available for 3,195 (99.7%) infants; 693 (21.7%) grew at least 1 organism. Organisms identified as significant pathogens were isolated in 564/693 (>1 pathogen in 29) blood cultures, contaminants (presumed non-pathogens) in 117, and indeterminate in 12 cultures (Table 1). Gram-negative and gram-positive pathogens were found in 62.9% (355/564) and 34.8% (196/564) of infants, respectively (n = 8 with both), and fungal pathogens in 21. Among infants with a significant pathogen, *Klebsiella pneumoniae* (23.4%, n = 132), Coagulase-negative staphylococci (CoNS) (14.9%, n = 84), *Acinetobacter* species (12.8%, n = 72), *S. aureus* (9.6%, n = 54), and *Escherichia coli* (8.3%, n = 47) were the most common (Table 1 and S20 Fig) but with differences between sites (S21 Fig). *Streptococcus agalactiae* was found in only 19 babies (3.4%). All the common pathogens were identified in both early and late onset sepsis (Fig 2), although *E. coli* was more common in the first 3 days of life while *K. pneumoniae* was more frequent in late onset sepsis (S8 Table).

Approximately 58% (75/130 tested) of *K. pneumoniae* isolates were resistant to gentamicin, 75.0% (96/128) to commonly used third-generation cephalosporins (cefotaxime/ceftriaxone),



Fig 2. Pathogens isolated in baseline blood culture, overall and by infants' day of life.

https://doi.org/10.1371/journal.pmed.1004179.g002





46.5% (53/114) to piperacillin-tazobactam, 46.6% (48/103) to ciprofloxacin, and 32.6% (43/132) to meropenem (S9 Table). *Acinetobacter* species were resistant to meropenem in 71.4% (50/70) of cases. *E. coli* retained greater susceptibility to third-generation cephalosporins (64.4%, 29/45 susceptible) and was susceptible in 90.4% (38/42) of cases to piperacillin-tazobactam. Among gram-negatives, there were important differences in susceptibility among aminoglycosides, with amikacin providing significantly better activity than gentamicin (e.g., 61.5% versus 38.5% susceptibility among *K. pneumoniae*) (Fig 3 and S9 Table).

Among 54 *S. aureus* isolates, 33 (61.1%) were methicillin resistant. *S. aureus* and CoNS were susceptible to vancomycin in all of 42 and 73 isolates tested, respectively, and all 19 *Streptococcus agalactiae* isolates were susceptible to ampicillin. Other rarer and more site-specific pathogens with high rates of resistance to antibiotics included *Serratia species*, *Burkholderia species*, and *Elizabethkingia meningoseptica* (S9 Table).

A total of 1,226/3,204 (38.3%) infants had a cerebrospinal fluid (CSF) culture performed at baseline or during the subsequent 7 days, and 73/1,226 (6.0%) were culture-positive, 47 with a pathogen, and 26 contaminant/indeterminate. Gram-negative organisms also dominated in CSF cultures (S10 Table).

# Mortality

Median follow-up time was 29 (IQR 28 to 29) days. Overall, 350 infants (11.3%; 95% CI 10.2% to 12.5%) died within 28 days of baseline blood culture. Mortality at 28 days was unknown in 62 infants (5 withdrew in hospital, 57 were lost post-discharge). There was significant variation in mortality between different sites (Fig 4A). Mortality among infants with a pathogen-positive baseline culture was 17.7% (99/564; 95% CI 14.7% to 21.1%) compared with 9.9% (250/2,631; 95% CI 8.8% to 11.2%) in infants without pathogens (p < 0.001). Mortality was higher in infants with gram-negative (21.3%; 95% CI 17.4% to 25.9%) or fungal pathogens (38.1%; 95% CI 21.2% to 61.9%) than with gram-positive pathogens (8.5%; 95% CI 5.3% to 13.5%; p < 0.001). Mortality was 33.3% (95% CI 23.7% to 45.5%) in infants with *Acinetobacter* spp., 21.5% (95% CI 15.4% to 29.6%) with *K. pneumoniae*, 21.1% (95% CI 8.5% to 46.8%) with *Streptococcus agalactiae*, 12.8% (95% CI 6.0% to 26.3%) with *E. coli*, 11.1% (95% CI 5.2% to 23.1%) with *S. aureus*, and 3.6% (95% CI 1.2% to 10.8%) with CoNS (Fig 4B and S11 Table). Overall, gram-negative infections accounted for 75/99 (75.8%) of culture-positive deaths, especially *K. pneumoniae* (n = 28; 28.3%) and *Acinetobacter* spp. (n = 24; 24.2%).

There was no significant association between time from blood culture to start of new antibiotics and mortality, neither overall (HR = 1.01 per additional 2 h [95% CI 0.96 to 1.07]; p = 0.60) nor in the subgroup with no previous antibiotic exposure (HR = 1.03 [95% CI 0.95 to 1.11]; p = 0.53) (S22 Fig).

### Baseline predictors of 28-day mortality

A prediction model was derived from 2,726 participants (n = 308 died) and validated in 478 (n = 42 died) (S12 and S13 Tables). Ten clinical factors known at presentation independently predicted mortality in the final model, including infant characteristics (birth weight, gestational age, duration of time in hospital, and congenital anomalies), level of respiratory support, and clinical signs (abnormal temperature, abdominal distension, lethargy/no or reduced movement, difficulty feeding, and evidence of shock) (Table 2). Risk was increased with both low ( $<35^{\circ}$ C) and high temperature ( $>38^{\circ}$ C) with evidence of even higher risk if  $>39^{\circ}$ C. The C-statistic for this model was 0.79 (95% CI: 0.77 to 0.81) and 0.81 (0.73 to 0.85) in the derivation and validation sample, respectively. A NeoSep Severity Score developed from these baseline predictors had a maximum 16 points (Table 2), with C-statistics 0.77 (95% CI: 0.75 to 0.80) and 0.76 (0.69 to 0.82) in the derivation and validation samples, respectively, and a good fit in the validation sample (Hosmer–Lemeshow p = 0.53). Defining mortality risk thresholds at 5% and 25% in the derivation sample categorized scores as low (score 0 to 4), medium (5 to 8), and high (9 to 16) risk, with mortality in the derivation sample of 1.6% (17/1,071; 95% CI: 1.0% to 2.5%), 14.4% (203/1,409; 12.7% to 16.3%), and 35.8% (88/246; 30.0% to 41.9%), and in the validation sample of 1.6% (3/189; 95% CI: 0.5% to 4.6%), 11.0% (27/245; 7.7% to 15.6%), and 27.3% (12/44; 16.3% to 41.8%) (Fig 5A).

The association between the NeoSep Severity Score and mortality was similar within multiple subgroups, for example, in early or late-onset, community or healthcare-associated, high-, middle- or low-income settings (Fig 5B and 5C), term versus preterm infants, blood culture-positive versus culture-negative cases (S23 Fig).

A score based on WHO pSBI criteria had a C-statistic of 0.63 (95% CI 0.56 to 0.70) in the validation sample, lower than the NeoSep Severity Score. Because the WHO pSBI criteria do not include any infant/birth characteristics, to ensure an appropriate comparison, we also



**Fig 4. Mortality.** (A) Overall mortality and mortality by site. (B) Mortality by pathogen in baseline blood culture. Figures present unadjusted Kaplan–Meier estimates.

https://doi.org/10.1371/journal.pmed.1004179.g004

Factor		NeoSep Severity Score (at presentation)				NeoSep Recovery Score (daily on IV antibiotics)			
		Coefficient (SE)	<i>p</i> -value	HR (95% CI)	Severity score points¶	Coefficient (SE)	<i>p</i> -value	HR (95% CI)	Recovery score points¶
Birth weight (kg) ‡	birth_wt^-2-0.181326	0.754 (0.162)	<0.001	1 kg: 2.13 (1.55, 2.92) 2 kg: 1.21 (1.12, 1.31) 3 kg: ref	<1 kg: 2 1-2 kg: 1				
Time in hospital	Per additional 24 h	-0.039 (0.011)	< 0.001	0.96 (0.94, 0.98)	$\leq$ 10 days: 1				
Gestational age	Per additional week	-0.064 (0.022)	0.004	0.94 (0.90, 0.98)	< 37 weeks: 1				
Congenital anomalies	Presence	1.032 (0.191)	< 0.001	2.81 (1.93, 4.08)	2				
Maximum respiratory support:	Oxygen supplementation CPAP, BiPAP, HFNC Invasive ventilation	1.055 (0.256) 1.551 (0.270) 2.318 (0.265)	<0.001 <0.001 <0.001	2.87 (1.74, 4.74) 4.72 (2.78, 8.00) 10.2 (6.05, 17.1)	2 3 3	2.221 (0.421) 2.721 (0.422) 3.883 (0.415)	<0.001 <0.001 <0.001	9.21 (4.04, 21.0) 15.2 (6.65, 34.8) 48.6 (21.5, 110)	2 3 3
Temperature (°C) ‡	Term 1: temperature^2– 1371.2072 term 2: temperature^2*ln (temperature)-4952.4209	$ \begin{array}{c} -0.740 \\ (0.240) \\ 0.181 (0.058) \end{array} $	0.002 0.002	35.5°C: 1.17 (0.78, 1.75) 37°C: ref 38°C: 1.41 (1.18, 1.69) 39°C: 2.91 (1.76, 4.81)	<35.5°C: 1 ≥38 -<39°C: 1 ≥39°C: 2	-0.839 (0.228) 0.205 (0.056)	<0.001 <0.001	35.5°C: 1.32 (0.97, 1.79) 37°C: ref 38°C: 1.43 (1.07, 1.92) 39°C: 3.07 (1.50, 6.31)	<35.5°C: 1 ≥38 -<39°C: 1 ≥39°C: 2
Abdominal distension	Presence	0.468 (0.131)	< 0.001	1.60 (1.24, 2.07)	1	0.407 (0.146)	0.005	1.50 (1.13, 2.00)	1
Lethargy, no or reduced movement:	Lethargy only No/reduced movement (± lethargy)	0.246 (0.142) 0.695 (0.198)	0.083 <0.001	1.28 (0.97, 1.69) 2.00 (1.36, 2.95)	1 2	0.617 (0.170) 1.130 (0.203)	<0.001 <0.001	1.85 (1.33, 2.59) 3.10 (2.08, 4.61)	1 2
Difficulty feeding	Presence	0.406 (0.141)	0.004	1.50 (1.14, 1.98)	1	0.763 (0.163)	< 0.001	2.15 (1.56, 2.95)	1
Evidence of shock	Presence	0.527 (0.143)	<0.001	1.69 (1.28, 2.24)	1	0.921 (0.167)	<0.001	2.51 (1.81, 3.48)	1
Cyanosis	Presence					0.624 (0.203)	0.002	1.87 (1.25, 2.78)	1

#### Table 2. Predictors of mortality and risk scores.

Maximum number of score points possible

This table shows results derived from 2 different multivariable Cox proportional hazards models with site-level random effects: (1) for a baseline NeoSep Severity Score to predict 28-day mortality from factors known at sepsis presentation; and (2) for a NeoSep Recovery Score to predict the daily risk of death while treated with IV antibiotics from daily updated assessments of clinical status.

16

\* Birth weight and temperature were analyzed as continuous variables using FPs with powers -2 for birth weight, and powers 2 2 for temperature. For illustrative reasons, in this table we report HRs for specific values.

<sup>9</sup> For each model, score points for each factor were derived by dividing its coefficient by the smallest significant coefficient of any categorical factor, rounding to the nearest integer, and giving 1 score points for <3 raw points, 2 score points for 3-5 raw points, and 3 score points for  $\geq 6$  raw points. Shaded factors: not a significant predictor (cyanosis: baseline severity score) or deliberately not included in model (unmodifiable factors excluded from recovery score: birth weight, time in hospital, gestational age, congenital anomalies).

BiPAP, Bilevel Positive Airway Pressure; CI, confidence interval; CPAP, Continuous Positive Airway Pressure; FP, fractional polynomial; HFNC, High Flow Nasal Cannula; HR, hazard ratio; SE, standard error.

https://doi.org/10.1371/journal.pmed.1004179.t002

11





calculated a modified NeoSep Severity Score excluding these factors that had a C-statistic of 0.72 (95% CI 0.65 to 0.77) (S24 Fig).

# Predictors of daily risk of dying of sepsis while on intravenous antibiotics in hospital: NeoSep Recovery Score

Seven of the time-varying factors independently predicted mortality in the final model, including all the non-modifiable signs in the NeoSep Severity Score (respiratory support and the 5 clinical signs), plus cyanosis, which had not added independent information in the baseline model (p = 0.228) (Table 2). A NeoSep Recovery Score developed from these time-updated predictors had a maximum 11 points and discriminated well between infants who died or survived the next day (Fig 6A and 6B). The AUROC of the recovery score on one day, for predicting death the next day, ranged between 0.8 and 0.9 over the first week post baseline in the derivation sample (Fig 7A). As expected, it slowly decreased over subsequent days. AUROC over time in the validation sample was similar although based on fewer numbers (Fig 7B).

The NeoSep Recovery Score on day 2 had an AUROC for dying in the following 5 days of 0.82 (95% CI 0.78 to 0.85) and 0.85 (95% CI 0.78 to 0.93) in the derivation and validation









samples, respectively. A score  $\geq$ 4 was the most discriminative, whether this was an increase from baseline, lack of initial response, or an improvement from a higher score down to 4, with sensitivity and specificity of 0.74 (95% CI 0.64 to 0.82) and 0.74 (0.72 to 0.75) in the derivation, and 0.87 (0.60 to 0.98) and 0.76 (0.71 to 0.79) in the validation samples.

Twenty-nine infants (derivation: n = 27; validation: n = 2) died between day 3 and 7 despite having had a score <4 on day 2. Of these, 6 deaths were not classified as infection-related, 10 had a subsequent increase in score to  $\geq$ 4 before dying, 9 had at least 2 unmodifiable risk factors included as predictors of mortality in the NeoSep Severity Score (most commonly congenital abnormalities and very low birth weight), and 1 had congenital varicella; the remaining 4 had a day 2 score of 2 or more.

Of note, the change in score from baseline to day 2 had poorer discrimination than the absolute score on day 2, with an AUROC for dying in the following 5 days of 0.68 (95% CI 0.63 to 0.73) and 0.62 (95% CI 0.48 to 0.77) in the derivation and validation samples, respectively. Combining change in score using various cut-offs with  $<4/\geq4$  absolute score on day 2 also did not improve discrimination (S26 Fig).

# Discussion

In this large hospital-based observational study in 11 countries across Africa, Asia, Europe, and Latin America, we observed wide variation in antibiotic prescribing beyond WHO recommendations for neonates and young infants with sepsis, with over 200 different empiric combinations, frequent empiric escalation, and rare de-escalation of therapy. Approximately 10% of pathogen-negative and 18% of pathogen-positive infants died. Gram-negative infections were associated with the majority of pathogen-positive deaths, and leading pathogens such as *K. pneumoniae, Acinetobacter* spp., and *S. aureus* were often resistant to WHO regimens. Mortality was independently associated with a range of factors, developed into a sepsis severity score which discriminated well between survivors and non-survivors at baseline. A recovery score was developed accounting for evolving clinical signs after baseline in order to assess response to antibiotics and also performed well at predicting death the following day. The NeoOBS study highlights the urgent need for clinical trials to inform antibiotic use for neonatal sepsis globally, describes the standard of care (SOC) against which novel antibiotic regimens should be compared, and provides severity and recovery scores that can be used for inclusion and antibiotic escalation criteria in future trial design.

Our data on antibiotic use are similar to recent point prevalence surveys in LMIC settings [16], but this study provides evidence on patterns of switching and escalation of therapy that is rarely available in the literature. A third of infants that had started on WHO-recommended regimens escalated to broader spectrum antibiotic regimens, and escalation overall was most commonly to a carbapenem. There was also considerable use of "carbapenem-sparing" regimens with "partial" ESBL and/or antipseudomonal activity that varied between centers (piper-acillin-tazobactam, ceftazidime, and quinolone-based regimens, often in combination with amikacin). Evidence on carbapenem-sparing regimens is limited in neonates [27], but *Pseudomonas* infections were rare in NeoOBS and in the recent published literature [5,6]. Notably, a small but important proportion of infants in some sites received treatment for proven or suspected carbapenem-resistant infection with colistin, for which a recommended approach to dosing and combination therapy remains unclear, and CSF penetration and side effect profile are suboptimal [28]. Some countries are also using less widely used combinations such as cefoperazone/sulbactam, for which neonatal data are limited [29,30]. There was almost no de-escalation of therapy.

AMR was common and gram-negative pathogens (Klebsiella and Acinetobacter) were largely resistant to WHO-recommended regimens. Among gram positives, MRSA accounted for over half of S. aureus isolates. However, the extent to which increasing prevalence of AMR is associated with excess mortality, and whether this may be modifiable with different antibiotic treatment strategies is unclear in neonates and young infants. AMR has been demonstrated as an independent predictor of mortality in adults with bloodstream infections, including in LMIC settings [31]. However, data in neonates and young infants in many LMIC settings are scarce [15,32]. A limited number of largely retrospective observational studies have shown an increase in mortality particularly in association with resistant gram-negative infections [33-36], such as those due to organisms producing ESBL [37-40] and CROs [41-44], and some suggest an increase in mortality in the absence of appropriate antimicrobial therapy [33,45-48]. A recent large multicenter neonatal sepsis study (BARNARDS) demonstrated high resistance to ampicillin + gentamicin (97% and 70%, respectively) among gramnegative infections; however, analysis of the influence of antibiotic treatment on outcomes was confounded by country effects, and limited clinical data prevented adjusting for other important confounders [9]. Appropriate analyses of the causal relationship between discordant antibiotic treatment and mortality would need to consider baseline confounders as well as timedependent confounding [49], but to our knowledge, this has not been done for neonatal sepsis in LMIC settings to an extent that could inform changes to global guidance [9,36,43,50,51]. Relevant analyses in the NeoOBS cohort are ongoing. Ultimately, evidence from randomized controlled trials is required to determine the most appropriate antibiotic regimens in different contexts.

The NeoSep Severity and Recovery Scores provided good discrimination. A score of 5 or higher at baseline was associated with 28-day mortality over 10%, and a recovery score 2 days after antibiotic initiation of 4 or higher had both sensitivity and specificity of 74% for mortality over the following 5 days. The NeoSep scores include clinical signs that have been designed to closely align with the WHO pSBI criteria (compared in S14 Table) used in recent communitybased studies, including AFRINEST [52] and SATT [53,54]. However, the NeoSep scores include additional factors more relevant to hospital settings which offer further predictive value, such as clinical evidence of shock, and need for oxygen and/or respiratory support. Importantly, the study population also included a more heterogenous mix of both preterm (including <1,500 g) and term infants, with early and late onset, and community and healthcare-associated sepsis, and the Severity and Recovery Scores performed similarly across all these subgroups. Of note, convulsions were uncommon in the study (7%) and were not associated with mortality despite being a pSBI criteria for critical illness, potentially due to low power. Previous severity scores derived from hospital-based LMIC cohorts have been based on smaller studies assessing general illness severity rather than sepsis specifically [55-60]. Highincome setting-based general illness severity scores are more widely used [55-57], and some have been assessed in small populations of septic neonates, but they are often unfeasible to apply globally [61,62]. The nSOFA score is a recent sepsis-specific score developed on 60 very low birth weight (<1,500 g) infants, but relies on measurement of parameters that are often unavailable in LMIC settings (e.g., use of inotropes/vasoactive drug use) [63]. The NeoSep scores are based on clinical variables that are relevant and feasible to collect globally across a range of LMIC hospital settings.

There are limitations to the generalizability of the data. The sites were selected to be heterogenous, varying in patient volume and case-mix, and with wide variation in levels of supportive care. Sites also varied in their ability to screen, using different screening schedules depending on staff levels and number of patients, particularly during nights or weekends. Given the very large numbers of infants presenting to many of the sites (see <u>S1 Table</u>), including out of hours, project funding was not available to prospectively screen and record the status of all admitted infants and reasons for not being enrolled. We would note, however, that the design enrolled participants who could feasibly be recruited to an antibiotic trial. Most sites in NeoOBS represented secondary or tertiary hospitals in urban settings, in some cases receiving referrals after prior treatment in other hospitals. In many cases, these facilities provided a higher level of care than is typical in many low-resource settings. AMR may be more important and a wider range of antibiotics may be available in some of the settings in this study compared to district hospital settings [64,65]. This is an important bias in most studies on AMR in low-resource settings, where the need for high-quality microbiology means certain settings are overrepresented.

A significant proportion (36.8%) of infants had previously received antibiotics before the new sepsis episode, which may influence culture yield and pathogen characteristics. Selection bias was also possible beyond site selection, with prospective recruitment potentially leading to a milder phenotype as infants who die rapidly with sepsis are more difficult to enroll, possibly leading to underestimation of mortality. There was wide variation in mortality between sites, highlighting major limitations of basing antibiotic recommendations on observational data alone. Analyses to determine the impact of antibiotic use on outcomes are confronted with important biases [49], including the influence of initial sepsis severity on antibiotic choice and timing of first administration, antibiotic availability and affordability, inter-site population heterogeneity, local guidelines and microbiology, and varying levels of supportive care.

In developing the NeoSep Severity and Recovery Scores, we attempted to account for intersite variation by using multivariable models with site-level random-effects, excluding factors with a high proportion of missing values. We also examined performance of the Severity Score across different sites in high-, middle-, and low-income contexts and demonstrated reproducible risk trends across all settings. Of note, the scores were validated on a predefined reserved 15% internal sample, including only 42 deaths (and relatively few infants with high-risk scores). External validation would strengthen the applicability of the scores.

In the context of increasing resistance to WHO-recommended therapy for neonatal sepsis in LMIC settings, and a lack of evidence to guide optimal management due to the limitations of observational data, further randomized antibiotic trials are urgently needed. This study has demonstrated that there is no single accepted "standard of care" in neonatal sepsis in LMIC hospital settings, and any novel regimens will need to be compared to multiple widely used empiric regimens including current WHO-recommended guidance. NeoOBS data, including the severity and recovery scores, have informed the design of the NeoSep1 trial (ISRCTN 48721236) which will use a network meta-analytic approach to rank novel off-patent antibiotic combinations compared to WHO-recommended and other commonly used regimens, combined with a SMART (Sequential Multiple Assessment Randomized Trial) design to allow randomization to both empiric first- and second-line treatment [66].

# Supporting information

S1 Appendix. Supplementary methods: Sample size determination. (PDF)
S1 STROBE checklist. STROBE checklist. (PDF)
S1 TRIPOD checklist. TRIPOD checklist. (DOCX)

**S1 Dataset. Data used to develop NeoSep Severity Score.** (XLSX)

S2 Dataset. Data used to develop NeoSep Recovery Score. (XLSX)

**S1 Fig. Clinical and laboratory sepsis enrolment criteria.** (PDF)

**S2 Fig. Study flow diagram.** (PDF)

**S3 Fig. Birth weight by site.** Boxes show 25th percentile (lower hinge), median (line), and 75th percentile (upper hinge); whiskers show lower and upper adjacent values as defined by Tukey. (TIF)

**S4 Fig. Hospitalized since birth vs. admitted, by site.** Proportion of infants hospitalized since birth and proportion admitted from home/community, per site. (TIF)

**S5 Fig. Postnatal age at enrolment by site.** Boxes show 25th percentile (lower hinge), median (line), and 75th percentile (upper hinge); whiskers show lower and upper adjacent values as defined by Tukey.

(TIF)

**S6 Fig. Frequency of supportive measures across sites.** This figure shows the number of sites (out of 19 overall) per percentage category of participants receiving a particular supportive measure at enrolment.

(TIF)

**S7 Fig. Prevalence of clinical signs/respiratory support over time.** Prevalence of clinical symptoms that are part of the NeoSep Severity Score and use of respiratory support over time in infants on IV antibiotics. (TIF)

[][])

**S8 Fig. Prevalence of clinical signs at enrolment (** $\leq$ **5%).** Prevalence of less common clinical symptoms at enrolment ( $\leq$ **5%).** (TIF)

**S9 Fig. Time to start IV antibiotics after baseline culture.** Time to first new IV antibiotic, by antibiotic exposure at enrolment. *P*-value derived from a log-rank test. (TIF)

**S10 Fig. Group 1 antibiotics.** Group 1: First-line WHO-recommended penicillin-based regimen (Access): Most common antibiotics used as initial regimen. (TIF)

**S11 Fig. Group 2 antibiotics.** Group 2: Third-generation cephalosporin-based WHO regimens ("Low" Watch), most common antibiotics used as initial regimen. (TIF)

**S12 Fig. Group 3 antibiotics.** Group 3: Regimens with partial anti-extended-spectrum betalactamase (ESBL) or pseudomonal activity ("Medium" Watch), most common antibiotics used as initial regimen. Pip/taz = piperacillin/tazobactam. (TIF) **S13 Fig. Group 4 antibiotics.** Group 4: Carbapenems ("High" Watch), most common antibiotics used as initial regimen.

(TIF)

**S14 Fig. Group 5 antibiotics.** Group 5: Reserve antibiotics targeting carbapenem resistant organisms (e.g., colistin), most common antibiotics used as initial regimen. (TIF)

**S15 Fig. "Other" group antibiotics.** Other antibiotics used as initial regimen. (TIF)

**S16** Fig. Antibiotic use on day 1 and on day 4 post baseline. (TIF)

**S17 Fig. Antibiotic use on day 4, by initial regimen.** (TIF)

**S18 Fig. Daily antibiotic regimens, by initial regimen and overall.** Cross-sectional analysis. Ignoring (unknown) treatment after transfer/readmission to another hospital. (TIF)

**S19 Fig. Days on antibiotics during follow-up.** Peak at days 27/28 due to infants still on IV antibiotics at the end of follow-up. (TIF)

**S20 Fig. Prevalence of pathogens isolated from baseline blood culture.** Prevalence of pathogens in participants with a baseline blood culture. (TIF)

**S21** Fig. Pathogens isolated from baseline blood culture, by site. (TIF)

**S22 Fig. Time to death, by time to start new IV antibiotics after baseline culture.** *P*-value derived from log-rank test.

(TIF)

**S23 Fig. NeoSep Severity Score performance in subgroups.** (A) Mortality (95% CI) in risk groups based on the Severity Score and pathogen/no pathogen at baseline. (B) Mortality (95% CI) in risk groups based on the Severity Score and preterm/term birth. (TIF)

**S24 Fig. Comparison between WHO pSBI signs and NeoSep Severity Score for predicting 28-day mortality.** Score based on WHO signs of Clinical Severe Infection, compared with a modified NeoSep Severity Score excluding unmodifiable infant/birth characteristics. pSBI = Possible serious bacterial infection. (PDF)

**S25 Fig. NeoSep Recovery Score over time (cross-sectional analysis).** Dashed line: for each score point, indicates those babies who eventually die on antibiotics; \* Restart of antibiotics ignored.

(TIF)

**S26 Fig. NeoSep Recovery Score on day 2 and change in score from baseline to day 2.** (A) Derivation data. (B) Validation data. (TIF)

**S1 Table. Candidate factors for the development of the NeoSep Severity Score.** \* Severe chest wall in-drawing, increased requirement for oxygen or respiratory support; T Observed or reported, including feeding intolerance.

(PDF)

**S2** Table. Numbers of births and admissions in each site (per 6 months). (PDF)

**S3 Table. Congenital anomalies.** Proportion of infants with congenital anomalies at enrolment.

(PDF)

**S4 Table. Laboratory results from blood at baseline.** IQR = interquartile range, CRP = C-reactive protein.

(PDF)

**S5 Table. Most common initial antibiotic regimens.** Most common initial regimens in infants who started new IV antibiotics within 24 h from baseline blood culture. (PDF)

S6 Table. Most common initial antibiotic regimens, by time from admission. HAI = healthcare-associated infection (occurring  $\geq$ 48 h after admission). (PDF)

**S7 Table. Factors associated with non-WHO-recommended regimens (Groups 3–5).** Note: OR = odds ratio, derived from logistic regression models adjusted for center. Time in hospital was analyzed as continuous variable using fractional polynomials with powers 1 1. For illustrative reasons, in this table we report odds ratios for specific values. (PDF)

**S8 Table. Organisms isolated from blood at baseline, by time from admission.** Note: \* Organisms isolated in only 1 infant per group. (PDF)

**S9 Table. Susceptibility result for pathogens in baseline blood culture.** Note: S = susceptible; I = intermediate; R = resistant. Denominators across drugs variables due to available susceptibility results. (PDF)

**S10 Table. Organisms isolated from CSF in first 7 days from baseline.** Note: *n* = 73 with positive CSF culture in first 7 days. (PDF)

**S11 Table. Pathogens and mortality.** Note: Results are hazard ratio (95% confidence interval). All models are adjusted for site (random effect). \* Adjusted for birth weight, gestational age, time in hospital, congenital anomalies, and site; \*\* adjusted for all factors in NeoSep Severity Score (as in \* plus abdominal distension, difficulty in feeding, evidence of shock, lethargy/no movement, temperature, and level of respiratory support); \*\*\* including Coagulase-negative staphylococcus. ref. = reference category. (PDF)

**S12 Table. Characteristics of participants in derivation and validation samples.** <sup>1</sup> Outcome for developing NeoSep Severity Score; <sup>2</sup> outcome for developing NeoSep Recovery Score. (PDF)

**S13 Table. Unadjusted analyses.** Notes: This table shows results of unadjusted Cox proportional hazards models for selected factors with site-level random effects: (1) for a baseline Neo-Sep Severity Score to predict 28-day mortality from factors known at sepsis presentation; and (2) for a NeoSep Recovery Score to predict the daily risk of death while treated with IV antibiotics from daily updated assessments of clinical status.<sup>‡</sup> Birth weight and temperature were analyzed as continuous variables using fractional polynomials with powers –2 for birth weight, and powers 2 2 for temperature. For illustrative reasons, in this table we report HRs for specific values. Shaded factors: deliberately not included in model (unmodifiable factors excluded from recovery score: birth weight, time in hospital, gestational age, congenital anomalies). CPAP = Continuous Positive Airway Pressure, BiPAP = Bilevel Positive Airway Pressure, HFNC = High Flow Nasal Cannula, HR = hazard ratio, CI = confidence interval, coef = coefficient.

(PDF)

**S14 Table.** Comparison of factors in NeoSep Sepsis Severity score with existing scores and sepsis criteria. CRIB = Clinical Risk Index for Babies Score; NMR score = Neonatal Mortality Risk score; SAWS = Simplified Age-Weight-Sex score; HR = Heart Rate; LBW = Low birth weight; SpO<sub>2</sub> = Oxygen Saturation. (PDF)

# Acknowledgments

This study would not have been possible without the infants and parents who took part in this trial. We also pay tribute to the commitment and contribution of all members of the clinical and research teams from all sites. We are grateful to the regional GARDP staff for their engagement with onsite monitoring, and we thank the PENTA foundation for their collaboration and role in study management and contractual agreements, and the University of Antwerp for microbiological oversight, quality assurance and analysis. We are indebted to national partners including ministries of health, participating hospitals and research institutions for hosting, facilitating and conducting the research, and we are grateful to the steering committee for their oversight and guidance throughout the study.

# **Author Contributions**

- **Conceptualization:** James A. Berkley, Manica Balasegaram, Sally Ellis, Julia A. Bielicki, A. Sarah Walker, Paul T. Heath, Mike Sharland.
- Data curation: Wolfgang Stöhr, Aislinn Cook, Amy Riddell.
- **Formal analysis:** Neal J. Russell, Wolfgang Stöhr, Aislinn Cook, Julia A. Bielicki, A. Sarah Walker, Mike Sharland.
- **Funding acquisition:** Manica Balasegaram, Sally Ellis, A. Sarah Walker, Paul T. Heath, Mike Sharland.
- Investigation: Neal J. Russell, Wolfgang Stöhr, Nishad Plakkal, Aislinn Cook, James A. Berkley, Bethou Adhisivam, Ramesh Agarwal, Nawshad Uddin Ahmed, Daynia Ballot, Adrie Bekker, Eitan Naaman Berezin, Suppawat Boonkasidecha, Cristina G. Carvalheiro, Neema Chami, Suman Chaurasia, Sara Chiurchiu, Viviane Rinaldi Favarin Colas, Tim R. Cressey, Ana Carolina Dantas de Assis, Tran Minh Dien, Yijun Ding, Nguyen Trong Dung, Han Dong, Angela Dramowski, Madhusudhan DS, Ajay Dudeja, Jinxing Feng, Youri Glupczynski, Srishti Goel, Herman Goossens, Doan Thi Huong Hao, Mahmudul Islam Khan,

Mohammad Shahidul Islam, Daniel Jarovsky, Meera Khorana, Angeliki Kontou, Tomislav Kostyanev, Premsak Laoyookhon, Sorasak Lochindarat, Mattias Larsson, Maia De Luca, Surbhi Malhotra-Kumar, Nivedita Mondal, Nitu Mundhra, Philippa Musoke, Marisa M. Mussi-Pinhata, Ruchi Nanavati, Firdose Nakwa, Sushma Nangia, Jolly Nankunda, Christina W. Obiero, Maxensia Owor, Wang Ping, Kanchana Preedisripipat, Lifeng Qi, Tanusha Ramdin, Amy Riddell, Lorenza Romani, Praewpan Roysuwan, Robin Saggers, Emmanuel Roilides, Samir K. Saha, Kosmas Sarafidis, Valerie Tusubira, Reenu Thomas, Sithembiso Velaphi, Tuba Vilken, Xiaojiao Wang, Yajuan Wang, Yonghong Yang, Liu Zunjie, Julia A. Bielicki, A. Sarah Walker, Paul T. Heath, Mike Sharland.

- **Methodology:** Neal J. Russell, Wolfgang Stöhr, Aislinn Cook, James A. Berkley, Ramesh Agarwal, Adrie Bekker, Neema Chami, Suman Chaurasia, Simon Cousens, Herman Goossens, Tomislav Kostyanev, Shamim Qazi, Amy Riddell, Samir K. Saha, Sithembiso Velaphi, Tuba Vilken, Sally Ellis, Julia A. Bielicki, A. Sarah Walker, Paul T. Heath, Mike Sharland.
- Project administration: Neal J. Russell, Wolfgang Stöhr, Nishad Plakkal, Aislinn Cook, James A. Berkley, Bethou Adhisivam, Ramesh Agarwal, Nawshad Uddin Ahmed, Manica Balasegaram, Daynia Ballot, Adrie Bekker, Eitan Naaman Berezin, Davide Bilardi, Suppawat Boonkasidecha, Cristina G. Carvalheiro, Neema Chami, Suman Chaurasia, Sara Chiurchiu, Viviane Rinaldi Favarin Colas, Tim R. Cressey, Ana Carolina Dantas de Assis, Tran Minh Dien, Yijun Ding, Nguyen Trong Dung, Han Dong, Angela Dramowski, Madhusudhan DS, Ajay Dudeja, Jinxing Feng, Youri Glupczynski, Srishti Goel, Doan Thi Huong Hao, Mahmudul Islam Khan, Tatiana Munera Huertas, Mohammad Shahidul Islam, Daniel Jarovsky, Nathalie Khavessian, Meera Khorana, Angeliki Kontou, Tomislav Kostyanev, Premsak Laoyookhon, Sorasak Lochindarat, Mattias Larsson, Maia De Luca, Nivedita Mondal, Nitu Mundhra, Philippa Musoke, Marisa M. Mussi-Pinhata, Ruchi Nanavati, Firdose Nakwa, Sushma Nangia, Jolly Nankunda, Alessandra Nardone, Borna Nyaoke, Christina W. Obiero, Maxensia Owor, Wang Ping, Kanchana Preedisripipat, Lifeng Qi, Tanusha Ramdin, Amy Riddell, Lorenza Romani, Praewpan Roysuwan, Robin Saggers, Emmanuel Roilides, Samir K. Saha, Kosmas Sarafidis, Valerie Tusubira, Reenu Thomas, Sithembiso Velaphi, Tuba Vilken, Xiaojiao Wang, Yajuan Wang, Yonghong Yang, Liu Zunjie, Julia A. Bielicki, A. Sarah Walker, Paul T. Heath, Mike Sharland.
- Supervision: Neal J. Russell, Wolfgang Stöhr, Aislinn Cook, James A. Berkley, Ramesh Agarwal, Simon Cousens, Tatiana Munera Huertas, Nathalie Khavessian, Tomislav Kostyanev, Surbhi Malhotra-Kumar, Shamim Qazi, Amy Riddell, Samir K. Saha, Sithembiso Velaphi, Tuba Vilken, Yonghong Yang, Sally Ellis, Julia A. Bielicki, A. Sarah Walker, Paul T. Heath, Mike Sharland.
- Validation: Neal J. Russell, Wolfgang Stöhr, Aislinn Cook, Tomislav Kostyanev, Tuba Vilken, A. Sarah Walker.
- Writing original draft: Neal J. Russell, Wolfgang Stöhr.
- Writing review & editing: Neal J. Russell, Wolfgang Stöhr, Nishad Plakkal, Aislinn Cook, James A. Berkley, Bethou Adhisivam, Ramesh Agarwal, Nawshad Uddin Ahmed, Manica Balasegaram, Daynia Ballot, Adrie Bekker, Eitan Naaman Berezin, Davide Bilardi, Suppawat Boonkasidecha, Cristina G. Carvalheiro, Neema Chami, Suman Chaurasia, Sara Chiurchiu, Viviane Rinaldi Favarin Colas, Simon Cousens, Tim R. Cressey, Ana Carolina Dantas de Assis, Tran Minh Dien, Yijun Ding, Nguyen Trong Dung, Han Dong, Angela Dramowski, Madhusudhan DS, Ajay Dudeja, Jinxing Feng, Youri Glupczynski, Srishti Goel, Herman Goossens, Doan Thi Huong Hao, Mahmudul Islam Khan, Tatiana Munera

Huertas, Mohammad Shahidul Islam, Daniel Jarovsky, Nathalie Khavessian, Meera Khorana, Angeliki Kontou, Tomislav Kostyanev, Premsak Laoyookhon, Sorasak Lochindarat, Mattias Larsson, Maia De Luca, Surbhi Malhotra-Kumar, Nivedita Mondal, Nitu Mundhra, Philippa Musoke, Marisa M. Mussi-Pinhata, Ruchi Nanavati, Firdose Nakwa, Sushma Nangia, Jolly Nankunda, Alessandra Nardone, Borna Nyaoke, Christina W. Obiero, Maxensia Owor, Wang Ping, Kanchana Preedisripipat, Shamim Qazi, Lifeng Qi, Tanusha Ramdin, Amy Riddell, Lorenza Romani, Praewpan Roysuwan, Robin Saggers, Emmanuel Roilides, Samir K. Saha, Kosmas Sarafidis, Valerie Tusubira, Reenu Thomas, Sithembiso Velaphi, Tuba Vilken, Xiaojiao Wang, Yajuan Wang, Yonghong Yang, Liu Zunjie, Sally Ellis, Julia A. Bielicki, A. Sarah Walker, Paul T. Heath, Mike Sharland.

### References

- Seale AC, Blencowe H, Manu AA, Nair H, Bahl R, Qazi SA, et al. Estimates of possible severe bacterial infection in neonates in sub-Saharan Africa, south Asia, and Latin America for 2012: a systematic review and meta-analysis. Lancet Infect Dis. 2014; 14:731–741. https://doi.org/10.1016/S1473-3099 (14)70804-7 PMID: 24974250
- Breiman RF, Blau DM, Mutevedzi P, Akelo V, Mandomando I, Ogbuanu IU, et al. Postmortem investigations and identification of multiple causes of child deaths: An analysis of findings from the Child Health and Mortality Prevention Surveillance (CHAMPS) network. PLoS Med. 2021; 18:e1003814. https://doi. org/10.1371/journal.pmed.1003814 PMID: 34591862
- Gabrysch S, Nesbitt RC, Schoeps A, Hurt L, Soremekun S, Edmond K, et al. Does facility birth reduce maternal and perinatal mortality in Brong Ahafo, Ghana? A secondary analysis using data on 119 244 pregnancies from two cluster-randomised controlled trials. Lancet Glob Health. 2019; 7:e1074–e1087. https://doi.org/10.1016/S2214-109X(19)30165-2 PMID: 31303295
- Fuchs A, Bielicki J, Mathur S, Sharland M, Van Den Anker JN. Reviewing the WHO guidelines for antibiotic use for sepsis in neonates and children. Paediatr Int Child Health. 2018; 38:S3–S15. <u>https://doi.org/ 10.1080/20469047.2017.1408738 PMID: 29790842</u>
- Okomo U, Akpalu ENK, Le Doare K, Roca A, Cousens S, Jarde A, et al. Aetiology of invasive bacterial infection and antimicrobial resistance in neonates in sub-Saharan Africa: a systematic review and metaanalysis in line with the STROBE-NI reporting guidelines. Lancet Infect Dis. 2019 [cited 2019 Sep 15]. https://doi.org/10.1016/S1473-3099(19)30414-1 PMID: 31522858
- Chaurasia S, Sivanandan S, Agarwal R, Ellis S, Sharland M, Sankar MJ. Neonatal sepsis in South Asia: huge burden and spiralling antimicrobial resistance. BMJ. 2019; 364:k5314. <u>https://doi.org/10.1136/ bmj.k5314 PMID: 30670451</u>
- Investigators of the Delhi Neonatal Infection Study (DeNIS) collaboration. Characterisation and antimicrobial resistance of sepsis pathogens in neonates born in tertiary care centres in Delhi, India: a cohort study. Lancet Glob Health. 2016; 4:e752–e760. https://doi.org/10.1016/S2214-109X(16)30148-6
  PMID: 27633433
- Sands K, Carvalho MJ, Portal E, Thomson K, Dyer C, Akpulu C, et al. Characterization of antimicrobialresistant Gram-negative bacteria that cause neonatal sepsis in seven low- and middle-income countries. Nat Microbiol. 2021; 6:512–523. https://doi.org/10.1038/s41564-021-00870-7 PMID: 33782558
- Thomson KM, Dyer C, Liu F, Sands K, Portal E, Carvalho MJ, et al. Effects of antibiotic resistance, drug target attainment, bacterial pathogenicity and virulence, and antibiotic access and affordability on outcomes in neonatal sepsis: an international microbiology and drug evaluation prospective substudy (BARNARDS). Lancet Infect Dis. 2021. <u>https://doi.org/10.1016/S1473-3099(21)00050-5</u> PMID: 34384533
- Murray CJL, Ikuta KS, Sharara F, Swetschinski L, Robles Aguilar G, Gray A, et al. Global burden of bacterial antimicrobial resistance in 2019: a systematic analysis. Lancet. 2022; 399:629–655. <u>https://doi.org/10.1016/S0140-6736(21)02724-0</u> PMID: 35065702
- Laxminarayan R, Matsoso P, Pant S, Brower C, Røttingen J-A, Klugman K, et al. Access to effective antimicrobials: a worldwide challenge. Lancet. 2016; 387:168–175. https://doi.org/10.1016/S0140-6736 (15)00474-2 PMID: 26603918
- Dewez JE, Chellani HK, Halim A, Van Den Broek N. Simplified antibiotic regimens for neonatal sepsis— AFRINEST(1). Lancet. 2015. https://doi.org/10.1016/S0140-6736(15)00330-X PMID: 26460772
- **13.** Mir F, Nisar I, Tikmani SS, Baloch B, Shakoor S, Jehan F, et al. Simplified antibiotic regimens for treatment of clinical severe infection in the outpatient setting when referral is not possible for young infants in Pakistan (Simplified Antibiotic Therapy Trial [SATT]): a randomised, open-label, equivalence trial.

Lancet Glob Health. 2017; 5:e177–e185. https://doi.org/10.1016/S2214-109X(16)30335-7 PMID: 27988146

- Joseph G, da Silva ICM, Wehrmeister FC, Barros AJD, Victora CG. Inequalities in the coverage of place of delivery and skilled birth attendance: analyses of cross-sectional surveys in 80 low and middleincome countries. Reprod Health. 2016; 13:77. <u>https://doi.org/10.1186/s12978-016-0192-2</u> PMID: 27316970
- Korang SK, Safi S, Nava C, Greisen G, Gupta M, Lausten-Thomsen U, et al. Antibiotic regimens for late-onset neonatal sepsis. Cochrane Database Syst Rev. 2021. https://doi.org/10.1002/14651858. CD013836.pub2 PMID: 33998665
- Jackson C, Hsia Y, Basmaci R, Bielicki J, Heath PT, Versporten A, et al. Global Divergence From World Health Organization Treatment Guidelines for Neonatal and Pediatric Sepsis. Pediatr Infect Dis J. 2019; 38. Available from: https://journals.lww.com/pidj/Fulltext/2019/11000/Global\_Divergence\_From\_ World\_Health\_Organization.12.aspx PMID: 31425329
- Folgori L, Ellis SJ, Bielicki JA, Heath PT, Sharland M, Balasegaram M. Tackling antimicrobial resistance in neonatal sepsis. Lancet Glob Health. 2017; 5:e1066–e1068. https://doi.org/10.1016/S2214-109X(17) 30362-5 PMID: 29025624
- Li G, Bielicki JA, Ahmed ASMNU, Islam MS, Berezin EN, Gallacci CB, et al. Towards understanding global patterns of antimicrobial use and resistance in neonatal sepsis: insights from the NeoAMR network. Arch Dis Child. 2020; 105:26 LP–31. https://doi.org/10.1136/archdischild-2019-316816 PMID: 31446393
- **19.** WHO. Managing possible serious bacterial infection in young infants when referral is not feasible. 2015. http://apps.who.int/iris/bitstream/handle/10665/181426/9789241509268\_eng.pdf?sequence=1.
- Tuzun F, Ozkan H, Cetinkaya M, Yucesoy E, Kurum O, Cebeci B, et al. Is European Medicines Agency (EMA) sepsis criteria accurate for neonatal sepsis diagnosis or do we need new criteria? PLoS ONE. 2019; 14:e0218002–e0218002. https://doi.org/10.1371/journal.pone.0218002 PMID: 31170237
- Fitchett EJA, Seale AC, Vergnano S, Sharland M, Heath PT, Saha SK, et al. Strengthening the Reporting of Observational Studies in Epidemiology for Newborn Infection (STROBE-NI): an extension of the STROBE statement for neonatal infection research. Lancet Infect Dis. 2016; 16:e202–e213. <a href="https://doi.org/10.1016/S1473-3099(16)30082-2">https://doi.org/10.1016/S1473-3099(16)30082-2</a> PMID: 27633910
- Collins GS, Reitsma JB, Altman DG, Moons KGM. Transparent reporting of a multivariable prediction model for individual prognosis or diagnosis (TRIPOD): the TRIPOD Statement. BMC Med. 2015; 13:1. https://doi.org/10.1186/s12916-014-0241-z PMID: 25563062
- Harris PA, Taylor R, Thielke R, Payne J, Gonzalez N, Conde JG. Research electronic data capture (REDCap)—A metadata-driven methodology and workflow process for providing translational research informatics support. J Biomed Inform. 2009; 42:377–381. <u>https://doi.org/10.1016/j.jbi.2008.08.010</u> PMID: 18929686
- 24. Hsia Y, Lee BR, Versporten A, Yang Y, Bielicki J, Jackson C, et al. Use of the WHO Access, Watch, and Reserve classification to define patterns of hospital antibiotic use (AWaRe): an analysis of paediatric survey data from 56 countries. Lancet Glob Health. 2019; 7:e861–e871. <u>https://doi.org/10.1016/S2214-109X(19)30071-3 PMID: 31200888</u>
- 25. Sulis G, Sayood S, Katukoori S, Bollam N, George I, Yaeger LH, et al. Exposure to WHO AWaRe antibiotics and isolation of multi-drug resistant bacteria: a systematic review and meta-analysis. Clin Microbiol Infect. 2022.
- Liang L, Kotadia N, English L, Kissoon N, Ansermino JM, Kabakyenga J, et al. Predictors of Mortality in Neonates and Infants Hospitalized With Sepsis or Serious Infections in Developing Countries: A Systematic Review. Front Pediatr. 2018: 277. Available from: <u>https://www.frontiersin.org/article/10.3389/ fped.2018.00277</u>.
- Lutsar I, Chazallon C, Trafojer U, de Cabre VM, Auriti C, Bertaina C, et al. Meropenem vs standard of care for treatment of neonatal late onset sepsis (NeoMero1): A randomised controlled trial. PLoS ONE. 2020; 15:e0229380. Available from: https://doi.org/10.1371/journal.pone.0229380 PMID: 32130261
- Nakwan N, Chokephaibulkit K, Imberti R. The Use of Colistin for the Treatment of Multidrug-resistant Gram-negative Infections in Neonates and Infants: A Review of the Literature. Pediatr Infect Dis J. 2019; 38. Available from: https://journals.lww.com/pidj/Fulltext/2019/11000/The\_Use\_of\_Colistin\_for\_ the\_Treatment\_of.13.aspx PMID: 31469781
- Ovali F, Gursoy T, Sari I, Divrikli D, Aktas A. Use of Cefoperazone/sulbactam in neonates. Pediatr Int. 2012; 54:60–63. https://doi.org/10.1111/j.1442-200X.2011.03458.x PMID: 21883691
- Bortone B, Jackson C, Hsia Y, Bielicki J, Magrini N, Sharland M. High global consumption of potentially inappropriate fixed dose combination antibiotics: Analysis of data from 75 countries. PLoS ONE. 2021; 16:e0241899. Available from: https://doi.org/10.1371/journal.pone.0241899 PMID: 33471786

- Stewardson AJ, Marimuthu K, Sengupta S, Allignol A, El-Bouseary M, Carvalho MJ, et al. Effect of carbapenem resistance on outcomes of bloodstream infection caused by Enterobacteriaceae in lowincome and middle-income countries (PANORAMA): a multinational prospective cohort study. Lancet Infect Dis. 2019; 19:601–610. https://doi.org/10.1016/S1473-3099(18)30792-8 PMID: 31047852
- 32. Wen SCH, Ezure Y, Rolley L, Spurling G, Lau CL, Riaz S, et al. Gram-negative neonatal sepsis in lowand lower-middle-income countries and WHO empirical antibiotic recommendations: A systematic review and meta-analysis. PLoS Med. 2021; 18:e1003787. Available from: https://doi.org/10.1371/ journal.pmed.1003787 PMID: 34582466
- 33. Peters L, Olson L, Khu DTK, Linnros S, Le NK, Hanberger H, et al. Multiple antibiotic resistance as a risk factor for mortality and prolonged hospital stay: A cohort study among neonatal intensive care patients with hospital-acquired infections caused by gram-negative bacteria in Vietnam. PLoS ONE. 2019; 14:e0215666–e0215666. https://doi.org/10.1371/journal.pone.0215666 PMID: 31067232
- 34. Folgori L, Livadiotti S, Carletti M, Bielicki J, Pontrelli G, Ciofi Degli Atti ML, et al. Epidemiology and Clinical Outcomes of Multidrug-resistant, Gram-negative Bloodstream Infections in a European Tertiary Pediatric Hospital During a 12-month Period. Pediatr Infect Dis J. 2014; 33. Available from: https://journals.lww.com/pidj/Fulltext/2014/09000/Epidemiology\_and\_Clinical\_Outcomes\_of.9.aspx.
- Foglia EE, Fraser VJ, Elward AM. Effect of Nosocomial Infections Due to Antibiotic-Resistant Organisms on Length of Stay and Mortality in the Pediatric Intensive Care Unit. Infect Control Hosp Epidemiol. 2007; 28:299–306. https://doi.org/10.1086/512628 PMID: 17326020
- Ivády B, Kenesei É, Tóth-Heyn P, Kertész G, Tárkányi K, Kassa C, et al. Factors influencing antimicrobial resistance and outcome of Gram-negative bloodstream infections in children. Infection. 2016; 44:309–321. https://doi.org/10.1007/s15010-015-0857-8 PMID: 26546372
- Abdel-Hady H, Hawas S, El-Daker M, El-Kady R. Extended-spectrum β-lactamase producing Klebsiella pneumoniae in neonatal intensive care unit. J Perinatol. 2008; 28:685. Available from: <u>https://doi.org/ 10.1038/jp.2008.73</u>.
- Kayange N, Kamugisha E, Mwizamholya DL, Jeremiah S, Mshana SE. Predictors of positive blood culture and deaths among neonates with suspected neonatal sepsis in a tertiary hospital, Mwanza-Tanzania. BMC Pediatr. 2010; 10:39. https://doi.org/10.1186/1471-2431-10-39 PMID: 20525358
- Ariffin H, Navaratnam P, Mohamed M, Arasu A, Abdullah WA, Lee CL, et al. Ceftazidime-resistant Klebsiella pneumoniae bloodstream infection in children with febrile neutropenia. Int J Infect Dis. 2000; 4:21–5. Available from: http://www.ncbi.nlm.nih.gov/pubmed/10689210 PMID: 10689210
- Zaoutis TE, Goyal M, Chu JH, Coffin SE, Bell LM, Nachamkin I, et al. Risk Factors for and Outcomes of Bloodstream Infection Caused by Extended-Spectrum ss-Lactamase-Producing Escherichia coli and Klebsiella Species in Children. Pediatrics. 2005; 115:942–949. <u>https://doi.org/10.1542/peds.2004-1289</u> PMID: 15805368
- 41. Thatrimontrichai A, Apisarnthanarak A, Chanvitan P, Janjindamai W, Dissaneevate S, Maneenil G. Risk Factors and Outcomes of Carbapenem-resistant Acinetobacter baumannii Bacteremia in Neonatal Intensive Care Unit: A Case-case-control Study. Pediatr Infect Dis J. 2013; 32. Available from: https:// journals.lww.com/pidj/Fulltext/2013/02000/Risk\_Factors\_and\_Outcomes\_of\_Carbapenem\_resistant. 10.aspx PMID: 22935872
- 42. Thatrimontrichai A, Techato C, Dissaneevate S, Janjindamai W, Maneenil G, Kritsaneepaiboon S, et al. Risk factors and outcomes of carbapenem-resistant Acinetobacter baumannii ventilator-associated pneumonia in the neonate: A case-case-control study. J Infect Chemother. 2016; 22:444–449. <u>https:// doi.org/10.1016/j.jiac.2016.03.013 PMID: 27229539</u>
- **43.** Punpanich W, Nithitamsakun N, Treeratweeraphong V, Suntarattiwong P. Risk factors for carbapenem non-susceptibility and mortality in Acinetobacter baumannii bacteremia in children. Int J Infect Dis. 2012; 16:e811–e815. https://doi.org/10.1016/j.ijid.2012.07.006 PMID: 22898148
- 44. Saleem AF, Ahmed I, Mir F, Ali SR, Zaidi AK. Pan-resistant Acinetobacter Infection in Neonates in Karachi, Pakistan. J Infect Dev Ctries. 2010; 4:030–037. https://doi.org/10.3855/jidc.533
- 45. Cook A, Hsia Y, Russell N, Sharland M, Cheung K, Grimwood K, et al. Association of Empiric Antibiotic Regimen Discordance With 30-Day Mortality in Neonatal and Pediatric Bloodstream Infection—A Global Retrospective Cohort Study. Pediatr Infect Dis J. 2021; 40. Available from: https://journals.lww. com/pidj/Fulltext/2021/02000/Association\_of\_Empiric\_Antibiotic\_Regimen.12.aspx PMID: 33395208
- Lenglet A, Faniyan O, Hopman J. A Nosocomial Outbreak of Clinical Sepsis in a Neonatal Care Unit (NCU) in Port-Au-Prince Haiti, July 2014—September 2015. PLoS Curr. 2018; 10. PMID: 29637010
- 47. Dramowski A, Aucamp M, Bekker A, Mehtar S. Infectious disease exposures and outbreaks at a South African neonatal unit with review of neonatal outbreak epidemiology in Africa. Int J Infect Dis. 2017; 57:79–85. https://doi.org/10.1016/j.ijid.2017.01.026 PMID: 28161461
- Naas T, Cuzon G, Robinson AL, Andrianirina Z, Imbert P, Ratsima E, et al. Neonatal infections with multidrug-resistant ESBL-producing E. cloacae and K. pneumoniae in Neonatal Units of two different

Hospitals in Antananarivo, Madagascar. BMC Infect Dis. 2016; 16:275. https://doi.org/10.1186/s12879-016-1580-5 PMID: 27287441

- 49. Lim C, Mo Y, Teparrukkul P, Hongsuwan M, Day NPJ, Limmathurotsakul D, et al. Effect of Delays in Concordant Antibiotic Treatment on Mortality in Patients With Hospital-Acquired Acinetobacter Species Bacteremia: Emulating a Target Randomized Trial With a 13-Year Retrospective Cohort. Am J Epidemiol. 2021; 190:2395–2404. https://doi.org/10.1093/aje/kwab158 PMID: 34048554
- 50. Dramowski A, Ong'ayo G, Rehman AM, Whitelaw A, Labi A-K, Obeng-Nkrumah N, et al. Mortality attributable to third-generation cephalosporin resistance in Gram-negative bloodstream infections in African hospitals: a multi-site retrospective study. JAC-Antimicrobial Resist. 2021; 3. <u>https://doi.org/10.1093/jacamr/dlaa130</u> PMID: 34223079
- Zaoutis TE, Goyal M, Chu JH, Coffin SE, Bell LM, Nachamkin I, et al. Risk factors for and outcomes of bloodstream infection caused by extended-spectrum beta-lactamase-producing Escherichia coli and Klebsiella species in children. Pediatrics. 2005; 115:942–9. https://doi.org/10.1542/peds.2004-1289 PMID: 15805368
- 52. Tshefu A, Lokangaka A, Ngaima S, Engmann C, Esamai F, Gisore P, et al. Oral amoxicillin compared with injectable procaine benzylpenicillin plus gentamicin for treatment of neonates and young infants with fast breathing when referral is not possible: a randomised, open-label, equivalence trial. Lancet. 2015; 385:1758–1766. https://doi.org/10.1016/S0140-6736(14)62285-6 PMID: 25842223
- 53. Baqui AH, Saha SK, Ahmed ASMNU, Shahidullah M, Quasem I, Roth DE, et al. Safety and efficacy of alternative antibiotic regimens compared with 7 day injectable procaine benzylpenicillin and gentamicin for outpatient treatment of neonates and young infants with clinical signs of severe infection when referral is not possible: a randomised, open-label, equivalence trial. Lancet Glob Health. 2015; 3:e279–e287. https://doi.org/10.1016/S2214-109X(14)70347-X PMID: 25841891
- Zaidi AKM, Tikmani SS, Warraich HJ, Darmstadt GL, Bhutta ZA, Sultana S, et al. Community-based treatment of serious bacterial infections in newborns and young infants: a randomized controlled trial assessing three antibiotic regimens. Pediatr Infect Dis J. 2012; 31:667–72. https://doi.org/10.1097/INF. 0b013e318256f86c PMID: 22481421
- 55. Aluvaala J, Collins GS, Maina M, Berkley JA, English M. A systematic review of neonatal treatment intensity scores and their potential application in low-resource setting hospitals for predicting mortality, morbidity and estimating resource use. Syst Rev. 2017; 6:248. <u>https://doi.org/10.1186/s13643-017-0649-6 PMID: 29212522</u>
- Dorling JS, Field DJ, Manktelow B. Neonatal disease severity scoring systems. Arch Dis Child Fetal Neonatal Ed. 2005; 90:F11 LP–F16. https://doi.org/10.1136/adc.2003.048488 PMID: 15613564
- Garg B, Sharma D, Farahbakhsh N. Assessment of sickness severity of illness in neonates: review of various neonatal illness scoring systems. J Matern Neonatal Med. 2018; 31:1373–1380. https://doi.org/ 10.1080/14767058.2017.1315665 PMID: 28372507
- 58. Medvedev MM, Brotherton H, Gai A, Tann C, Gale C, Waiswa P, et al. Development and validation of a simplified score to predict neonatal mortality risk among neonates weighing 2000 g or less (NMR-2000): an analysis using data from the UK and The Gambia. Lancet Child Adolesc Heal. 2020; 4:299–311. https://doi.org/10.1016/S2352-4642(20)30021-3 PMID: 32119841
- Rosenberg RE, Ahmed S, Saha SK, Ahmed ASMNU, Chowdhury MAKA, Law PA, et al. Simplified Age-Weight Mortality Risk Classification for Very Low Birth Weight Infants in Low-Resource Settings. J Pediatr. 2008; 153:519–524.e3. https://doi.org/10.1016/j.jpeds.2008.04.051 PMID: 18539298
- Mansoor KP, Ravikiran SR, Kulkarni V, Baliga K, Rao S, Bhat KG, et al. Modified Sick Neonatal Score (MSNS): A Novel Neonatal Disease Severity Scoring System for Resource-Limited Settings. Leenen PHL, editor. Crit Care Res Prac. 2019; 2019:9059073. https://doi.org/10.1155/2019/9059073 PMID: 31210987
- Rosenberg RE, Ahmed ASMNU, Saha SK, Chowdhury MAKA, Ahmed S, Law PA, et al. Nosocomial sepsis risk score for preterm infants in low-resource settings. J Trop Pediatr. 2010; 56:82–89. https:// doi.org/10.1093/tropej/fmp061 PMID: 19622712
- Pal S, Jain A, Garg M, Sekhar JC. Predicting Outcome in Neonates with Possible Clinical Sepsis by Estimating an Early Score for Neonatal Acute Physiology-II (SNAP-II). J Trop Pediatr. 2020; 66:377– 384. https://doi.org/10.1093/tropej/fmz076 PMID: 31682271
- Wynn JL, Polin RA. A neonatal sequential organ failure assessment score predicts mortality to lateonset sepsis in preterm very low birth weight infants. Pediatr Res. 2020; 88:85–90. <u>https://doi.org/10.1038/s41390-019-0517-2</u> PMID: 31394566
- Panigrahi P, Chandel DS, Hansen NI, Sharma N, Kandefer S, Parida S, et al. Neonatal sepsis in rural India: timing, microbiology and antibiotic resistance in a population-based prospective study in the community setting. J Perinatol. 2017; 37:911–921. https://doi.org/10.1038/jp.2017.67 PMID: 28492525

- 65. Saha SK, Schrag SJ, El Arifeen S, Mullany LC, Shahidul Islam M, Shang N, et al. Causes and incidence of community-acquired serious infections among young children in south Asia (ANISA): an observational cohort study. Lancet. 2018; 392:145–159. https://doi.org/10.1016/S0140-6736(18)31127-9 PMID: 30025808
- **66.** Walker AS, White IR, Turner RM, Hsu LY, Yeo TW, White NJ, et al. Personalised randomised controlled trial designs—a new paradigm to define optimal treatments for carbapenem-resistant infections. Lancet Infect Dis. 2021; 21:e175–e181. https://doi.org/10.1016/S1473-3099(20)30791-X PMID: 33894130